

Tsalavoutas, Y. and Tsoligkas, F. (2021) Uncertainty avoidance and stock price informativeness of future earnings. *Journal of International Financial Markets, Institutions and Money*, 75, 101410. (doi: 10.1016/j.intfin.2021.101410)

The material cannot be used for any other purpose without further permission of the publisher and is for private use only.

There may be differences between this version and the published version. You are advised to consult the publisher's version if you wish to cite from it.

http://eprints.gla.ac.uk/249671/

Deposited on 17 August 2021

Enlighten – Research publications by members of the University of Glasgow

http://eprints.gla.ac.uk

# Uncertainty avoidance and stock price informativeness of future earnings

Ioannis Tsalavoutas*
Fanis Tsoligkas
26 July 2021
20 July 2021
Acknowledgements
We are grateful for constructive and helpful comments on earlier versions of this work from two anonymous reviewers, Paul André, Diogenis Baboukardos, Richard Jackson, Alain Schatt, Christian Stadler, Antonios Siganos, Maximilian Wirth, and participants at the 23 <sup>rd</sup> Financial Reporting and Business Communication Conference (July 2019, Reading, UK) and the 2018 International Symposium of Finance (Chania, Greece).
Declarations of interest: none
* Corresponding Author
Ioannis Tsalavoutas is at the University of Glasgow, Adam Smith Business School, West Quadrangle, Main Building, Room G683, University Avenue, Glasgow, G12 8QQ, Scotland, UK. E-mail:

Fanis Tsoligkas is at the University of Bath, School of Management, Wessex House, Room 9.52, Claverton Down, Bath, BA2 7AY, UK. E-mail: F.Tsoligkas@bath.ac.uk.

ioannis.tsalavoutas@glasgow.ac.uk.

# Highlights

- We examine the effect of uncertainty avoidance on the share price informativeness fo future earnings.
- We conduct our analysis using an international sample of publicly listed firms.
- Uncertainty avoidance is negatively related to share price informativeness.
- Uncertainty avoidance affects the market's ability to anticipate future earnings.
- This relation is less pronounced within markets with higher market openness.

## Uncertainty avoidance and stock price informativeness of future earnings

#### **Abstract**

We explore whether uncertainty avoidance, an important aspect of national culture, influences the level of informativeness of share prices about future earnings. Uncertainty avoidance relates to the extent to which the members of a society feel threatened by uncertain or unknown situations. We employ data from 20 countries, comprising of 26,882 firm-year observations reporting under the same accounting standards. Using panel data analysis with OLS regressions, we show that firms' current stock returns incorporate less future earnings information in countries with high uncertainty avoidance. Further, we report that this relation is less pronounced within countries with relatively high market openness, consistent with the premise that the effect of national culture is reduced when a country's market is more open to foreign investors. Our study contributes to the literature by showing that international investors' innate differences affect their ability to anticipate future earnings and impound this information into current prices. As such, the benefits of adopting a common set of accounting standards are uneven across countries not only because of the way the standards are applied but also due to investors' innate differences.

*Keywords*: National Culture; Uncertainty Avoidance; Stock Market Efficiency; Stock Price Informativeness; Market Openness.

JEL classifications: G14; M41; M14

#### 1. Introduction

A strong relation between current stock returns and future firm earnings indicates an informationally efficient market. More informationally efficient prices facilitate more efficient resource allocation, because they reflect greater information and better capture the firm's underlying value (Durnev et al., 2003). Prior studies have shown that national culture influences several areas of business activities and decision making, including multinational corporation decentralization activities (Williams & van Triest, 2009), project evaluation (Harrison & McKinnon, 1999), organizational design, management planning and control systems (Harrison et al., 1994), compensation practices (Schuler & Rogovsky, 1998), dividend policy (Shao et al., 2010), marketing (Soares et al., 2007), asset managers' behavior (Beckmann et al., 2008), central banks' transparency (Makrychoriti & Pasiouras, 2021) and commercial banks' relationships with companies (Pasiouras et al., 2020) and risk taking (Kanagaretnam et al., 2014). However, only in more recent years has there been some interest in the significance of national culture on investor decisions. In line with this, Dou et al. (2016, p. 851) call for research that includes cultural dimensions '...in cross-country research to account for innate differences among international investors'. In this study, we respond to this call by examining whether uncertainty avoidance has an effect upon the stock price informativeness about future earnings as measured by the association between current returns and future earnings. This allows us to examine whether one aspect of culture is a fundamental factor that affects the process with which stock prices are formed.

Uncertainty avoidance is defined as "the degree to which the members of a society feel uncomfortable with uncertainty and ambiguity" (Hofstede, 1984, p.83). Thus, uncertainty avoidance captures the extent to which individuals tolerate uncertainty and feel threatened by ambiguous situations. Given this, we hypothesize that investors in high uncertainty avoidance societies are more likely to be uncomfortable with the level of uncertainty associated with

firms' future performance. Thus, stock prices in countries with cultures with higher uncertainty avoidance would exhibit a lower anticipation of future earnings (i.e., the association of current returns and future earnings will be lower in such markets). Subsequently, we corroborate our findings of this hypothesis by examining the effect of market openness. The participation of foreign investors introduces a multitude of norms and behaviours of foreign cultures and societies which then weakens the influence of domestic culture (Cowen, 2002; Jones, 2009; Eun et al., 2015).

We use data from 2005 to 2018 and our sample comprises of 26,882 firm-year observations from 20 countries. Specifically, the analysis focuses on firms reporting under International Financial Reporting Standards (IFRS) on a mandatory basis. This ensures that accounting numbers are more comparable among firms and across countries. We find that firms' current stock returns incorporate less future earnings information in countries with high uncertainty avoidance. In fact, this effect is also economically significant. Further, we find that the adverse effect of uncertainty avoidance remains robust to the inclusion of a number of firm and country controls which may affect the ability of stock returns to incorporate more future firm earnings information. This result confirms the main hypothesis. Additionally, we find that these results are driven by markets with relatively low market openness, where the market is mostly dominated by domestic investors. Finally, we find that our results also hold when we use aggregate country-level data, providing further support that our results capture a pervasive phenomenon (see also Kothari et al., 2006).

We contribute to the literature in the following ways. First, to the best of our knowledge, this is the first study establishing that national culture is indeed a determinant of share price

\_

<sup>&</sup>lt;sup>1</sup> Mandatory transition toIFRS has in effect caused a structural change by affecting the earnings-return relation (e.g. Choi, et al., 2013) resulting in increased comparability (Brochet et al., 2013; Neel, 2017; Yip & Young, 2012). Prior literature documents that comparability has a strong effect on the informativeness of stock prices (Choi et al., 2019).

informativeness about future earnings. Given that no cultural aspect has been considered thus far by this strand of the literature (e.g., Ettredge et al., 2005; Choi et al., 2011; Haw et al., 2012), this finding adds a new piece to the puzzle relating to the process of "stock price formation" (Haw et al., 2012 p. 391). Second, our study builds on and complements the studies by Chui et al. (2010) and Dou et al., (2016). The former explores the significance of one other element of national culture (individualism) in relation to the profitability of a stock market anomaly (the momentum effect). Dou et al., (2016) show that investors embedded in cultures with higher uncertainty avoidance (individualism) exhibit a lower (higher) under-reaction to earnings information. Our study also advances the conjectures in Maher and Parikh (2011) who focus on India and the findings by Papanastasopoulos (2014) who focuses on countries in the European Union. Both studies point to uncertainty avoidance as a potential influential factor of investors' behaviour. We show that uncertainty avoidance also influences the ability of investors to anticipate future earnings and impound this information into current prices, while market openness moderates this effect.

The evidence provided in this study also sheds light on the concerns raised in the literature about the potentially uneven benefits arising from the application of IFRS in different countries around the world because of differing country characteristics (Ball, 2006; Nobes, 2006; Weetman, 2006; Zeff, 2007). This gives rise to policy implications. The mission statement of the International Accounting Standards Board (IASB) describes IFRS as standards "[which] contribute to economic efficiency ... thus improving capital allocation". More informationally efficient prices facilitate more efficient resource allocation because they reflect greater information and better capture the firm's underlying value (Durnev et al., 2003). Given that

-

<sup>&</sup>lt;sup>2</sup> However, a number of factors may influence the profitability of a trading strategy such as the limits to arbitrage which are hard to control for (e.g., Stambaugh et al., 2012).

<sup>&</sup>lt;sup>3</sup> https://www.ifrs.org/about-us/

our sample comprises firms that adopted IFRS on a mandatory basis, our results show that the benefits of adopting a common set of accounting standards are indeed uneven across countries. Specifically, countries with higher levels of uncertainty avoidance exhibit lower levels of information efficiency. While IFRS' adoption may improve capital allocation it is not the only necessary and sufficient condition for information efficiency.

The remainder of the paper is structured as follows. Section 2 briefly reviews related literature and outlines our hypotheses. Section 3 presents the research design. Section 4 discusses the empirical results, including a battery of sensitivity tests we have conducted. Finally, section 5 concludes this study.

# 2. Relevant literature and hypotheses development

## 2.1 Price informativeness about future earnings

The stock return of a firm over the course of one year is partly due to, first, the realization of the unexpected portion of the current year's earnings and, second, the changes in expectations about future earnings (Collins et al., 1994). The latter implies that current stock returns reflect, to an extent, investors' expectations about forthcoming firm earnings (Warfield & Wild, 1992; Durnev et al., 2003). As Tucker and Zarowin (2006) explain, a key advantage of considering the anticipation of future earnings as reflected in current returns is that the change in expected future earnings may be due to a shock that has no effect on current earnings. While such information will be impounded in current stock prices, it may not be captured by current earnings. This insight has provided a fruitful area of research with a significant body of literature focusing on investors' ability to anticipate future firm earnings. As noted earlier, if more information about fundamentals and a company's future prospects is reflected in current stock returns, prices exhibit higher informativeness and informational efficiency, which facilitates more efficient resource allocation (Durnev et al., 2003).

Prior studies exploring the relation between current stock returns and future earnings have typically explored firm characteristics that influence the magnitude of this positive relation, mostly within the US market. For example, it has been reported that a firm's size and its earnings volatility are related with price informativeness: large-sized firms, and firms that smooth their earnings, tend to be more informationally efficient (Tucker & Zarowin, 2006). Prior literature also shows that financial statements seem to matter more when companies exhibit higher comparability, better earnings quality and within countries with stronger investor protection, better enforcement of insider trading laws and better financial disclosure through media (e.g., Hope, 2003; Bushman et al., 2004; Choi et al., 2011; Haw et al., 2012; Choi et al., 2013). However, this strand of the literature on price informativeness about future earnings has yet to consider the significance of national culture, and in particular the dimension of uncertainty avoidance.

2.2 Hypotheses development: Uncertainty avoidance as a determinant of firms' price informativeness about future earnings

Culture is particularly important in shaping 'perceptions, preferences, and behaviors and, therefore, action outcomes by and perceived utilities of the financial decision maker(s)' (Aggarwal et al., 2016, p. 467). Thus, culture is expected to have an important impact on investors' decisions and interactions with others. Individuals' upbringing and later experiences are expected to be more homogeneous within a country, and these experiences may well differ significantly among countries. Consistent with this premise, a number of recent studies have reported that particular cultural characteristics explain cross country differences in stock returns. Chui et al. (2010) is one of the first studies in the field of finance showing that individualism is related with cross-country momentum stock returns. Eun et al. (2015) report that individualism influences the co-movement of cross-country stock returns. Dou et al. (2016)

find that investor individualism and uncertainty avoidance are related with cross-country earnings momentum. Cheon and Lee (2018) highlight the significance of individualism in the cross-country differences in stock returns. Gaganis et al. (2021) show that firms from countries with higher secrecy, a composite measure from Hoftstede national culture measures, exhibit high stock price synchronicity. Finally, using a sample of cross listed firms in the US, Abdallah et al. (2021) show that stock price synchronicity is reduced following the adoption of IFRS and this effect is more pronounced for firms from countries which higher secrecy.

The definition of uncertainty avoidance, as provided by Hofstede (1984), indicates that it captures the extent to which individuals tolerate uncertainty and feel threatened by ambiguous situations.<sup>4</sup> Individuals in societies with high uncertainty avoidance feel more anxious or threatened in uncertain or ambiguous situations, while on the other hand, people in low uncertainty avoidance cultures tend to be more comfortable with uncertainty and ambiguity. Thus, individuals exhibiting high uncertainty avoidance experience uneasiness and reluctance when facing situations which increase uncertainty and ambiguity, and consequently attempt to avoid unknown and unfamiliar situations.

Consistently, prior management literature shows that a country's level uncertainty avoidance may discourage firms from engaging in business activities associated with high uncertainty such as the method of entry into a market (Png et al., 2001), adoption of information technology (IT) infrastructure (Kogut & Singh, 1988), supply chain collaborations (Qu & Yang, 2015) and cross-border/cross-industry takeovers (Frijns et al., 2013). Similarly, Abdolmohammadi and Sarens (2011) show that uncertainty avoidance is correlated with a lower usage and compliance of auditing standards and conclude that "standards that introduce

<sup>&</sup>lt;sup>4</sup> It could be argued that uncertainty avoidance is also related with risk aversion. However, risk relates to known probabilities. Further, Rieger et al. (2015) argue that although risk aversion depends on cultural characteristics such as uncertainty avoidance, 'equalizing... [uncertainty avoidance] with risk aversion is an oversimplification' (p. 639).

significant ambiguity/uncertainty in terms of interpretation and application may be subject to low usage" (p. 386).

In line with this, it has been argued that uncertainty avoidance could also influence investment decisions and deter individuals from investing in riskier assets. In fact, Beugelsdijk and Frijns (2010) and Anderson et al. (2011) show that investors from high uncertainty avoidance countries invest less in foreign assets which are considered riskier than domestic. Maher and Parikh (2011) use India as a country case study and document increased volatility prior to periods of high uncertainty, with a subsequent price correction as the uncertainty subsides. They interpret their findings by reflecting on India's specific cultural background, specifically pointing to uncertainty avoidance as defined by Hofstede. Papanastasopoulos (2014) employs firms from the European Union and shows that the accrual anomaly is stronger in countries with higher individualism and lower uncertainty avoidance. Building on these findings, Moreira et al. (2019) use a sample of firms from Latin America countries and show that the accrual anomaly is stronger in countries with a lower 'rule preference culture', which is defined as the sum of individualism and uncertainty avoidance. Moreover, prior literature documents that managers in countries with higher values of uncertainty avoidance would tend to hold more liquid assets such as cash (e.g., Chang & Noorbakhsh, 2009; Ramirez & Tadesse, 2009; Chen Y. et al., 2015). This is because individuals in such contexts "would tend to be less tolerant for uncertainty associated with future cash-flows generated by firms, and thus hold more cash to compensate for bearing this uncertainty" (Chen Y. et al., 2015, p. 4). The tolerance for uncertainty over the future in societies with higher uncertainty avoidance is consistent with an emphasis on "short-run reaction to short-run feedback rather than anticipation of long-run uncertainty. Such people solve pressing problems rather than develop long-run strategies" (Dou et al., 2016, p. 857).

The evidence in these studies suggests that investors in high uncertainty avoidance societies are more likely to be uncomfortable with the level of uncertainty associated with firms' future performance levels and the ambiguity related to the probability realisation of any forecasted future earnings. In fact, it is known that investors who receive ambiguous information are cautious and respond conservatively, by assuming the worst-case probability distribution (Epstein & Schneider, 2008; Williams, 2014) and hence discount ambiguous information (Van Dijk & Zeelenberg, 2003). Investors' incorporation of future earnings in stock prices in such environments would also be weaker because earnings management is known to be relatively high in countries with high uncertainty avoidance scores (Nabar & Boonlert-U-Thai, 2007). We thus conjecture that the combination of the financial reporting environment and investors' own response to uncertainty and ambiguity would then lead to current returns reflect less forward-looking earnings' information in cultures characterized by greater uncertainty avoidance. Effectively, future earnings will have less weight in current firm returns in such environments. Hence, we test the following hypothesis:

Hypothesis 1: The relation between current stock returns and firm future earnings is expected to be stronger (weaker) within countries characterised by lower (higher) uncertainty avoidance.

The underlying assumption of the first hypothesis is that investors' cultural attributes explain a weaker association between stock return and future earnings in countries with high uncertainty avoidance. However, arguably, the lower anticipation of future earnings in countries with high uncertainty avoidance could be driven by managers' cultural attributes inducing a lower quality of information. In order to corroborate our first hypothesis regarding the effect of culture on investors' anticipation of future earnings, we examine the effect of market openness as a moderator variable in this relationship.

As a result of economies' globalization, investors invest in international markets (e.g., Poole, 2004). The globalization of the economy allows for the transmission of local cultures to foreign countries and hence the presence of foreign investors (i.e., the degree of market openness) introduces cultural influences and norms of foreign societies. This results in the weakening/dilution of the effect of the local norms and domestic culture on overall investors' decisions in the long-run (Cowen, 2002; Jones, 2009). As Eun et al., (2015, p. 286) summarise this "the trading activities of foreign investors in domestic markets are likely to mitigate the influence of national culture on stock price behaviors".

Consistent with this idea, Stulz and Williamson (2003) show that market openness moderates the effect of religion on creditor rights. Eun et al. (2015) document that market openness moderates the effect of national culture, and individualism in particular, on the comovement of cross-country stock returns. Additionally, El Ghoul and Zheng (2016) show that market openness weakens the relation between domestic national culture and credit provision.

This evidence suggests that the hypothesized significant negative effect of uncertainty avoidance on the price informativeness of future earnings is likely to be affected (i.e., moderated) by the level of a country's market openness. We expect that the adverse effect of uncertainty avoidance is more prominent when relatively more local investors participate in the domestic stock market. Instead, when more international investors participate in the market, the relation should become weaker, since a large number of investors with varying cultures transact. We, thus, test the following hypothesis.

Hypothesis 2: The influence of uncertainty avoidance on the relation between current stock returns and firm future earnings is expected to be stronger (weaker) within countries with low (high) market openness.

### 3. Research design

#### 3.1 Methodology

In order to test our hypotheses, we follow the model originally developed by Collins et al. (1994) and later modified by Lundholm and Myers (2002) to measure the market's ability to anticipate future earnings. This model, which has extensively been used in prior literature (Gelb & Zarowin, 2002; Tucker & Zarowin, 2006; Hussainey & Walker, 2009; Haw et al., 2012; Chou, 2013; Dargenidou et al., 2021),<sup>5</sup> expresses current returns as a function of future, current and past earnings as follows:

$$R_{i,t} = a_0 + b_1 E_{i,t+1,+3} + b_2 E_{i,t} + b_3 E_{i,t-1} + b_4 R_{i,t+1,+3} + \varepsilon_{i,t}$$
 (1)

where  $R_{i,t}$  is the annual stock return of firm i in year t, measured from nine months before the year end to three months after the year end;  $E_{i,t+1,+3}$  is the sum of future earnings of firm i for the three years following the current year, scaled by the market value of equity measured at the firm's year end<sup>6</sup>; and  $E_{i,t}$  is the earnings of firm i and year t scaled by the market value of equity measured at the firm's year end. The model also includes returns  $R_{i,t+1,+3}$  as a control variable, because using actual future earnings to explain current returns introduces measurement error generated by events in future periods not anticipated in the current period.

The appealing feature of the model is that the coefficient of  $E_{i,t+1,+3}$ ,  $b_1$ , captures the market response to information about future earnings that is anticipated and not reflected in/captured by current and past earnings. This coefficient is predicted to be positive. The coefficient of  $E_{i,t-1}$ ,  $b_3$ , captures the already anticipated portion of current earnings and is expected to be negative. The coefficient of  $E_{i,t}$ ,  $b_2$ , represents the market response to the unexpected portion of current earnings and is predicted to be positive. Lastly, the coefficient of future returns,  $b_4$ , is expected

<sup>&</sup>lt;sup>5</sup> An alternative method to measure the information content of stock markets is the stock price synchronicity (e.g., Nguyen and Truong, 2013; Kim et al., 2018). However, we refrain from using this measure given the concerns discussed in the literature with respect to measurement bias in synchronicity proxies (e.g., Gassen et al., 2020).

<sup>&</sup>lt;sup>6</sup> We use three years of future earnings because Collins et al. (1994) and Lundholm and Myers (2002) show that considering more years adds little explanatory power.

to be negative, ensuring a 'better approximation to the changes in expectations of future earnings' (Collins et al., 1994, p. 299).

To test our first hypothesis, we extend Equation (1) by including our measure of uncertainty avoidance, UA, both as a main effect and as an interaction with future earnings,  $E_{i,t+1,+3}$ , as follows:

$$R_{i,t} = a_0 + b_1 E_{i,t+1,+3} + b_2 E_{i,t} + b_3 E_{i,t-1} + b_4 R_{i,t+1,+3} + b_5 E_{i,t+1,+3} * UA + b_6 UA + Controls +$$

$$Industry FE + Year FE + \varepsilon_{i,t}$$
(2)

where, uncertainty avoidance is measured using the relevant index from the Globe project (House et al., 2004). In this case, the coefficient of  $E_{i,t+1,+3}$  captures the association between current returns and future earnings for firms from countries with low uncertainty avoidance. The coefficient  $b_5$ , which is the coefficient of interaction between future earnings and uncertainty avoidance,  $E_{i,t+1,+3}*UA$ , captures the incremental effect of UA upon the association between future earnings and current returns. In line with hypothesis H1, we expect the coefficient of  $E_{i,t+1,+3}*UA$ ,  $b_5$ , to be negative and significant. This would indicate that the market's ability to anticipate future earnings is weaker for firms in countries with higher uncertainty avoidance reflecting the tolerance for uncertainty over the future.

In addition, we include a battery of firm-level control variables which may affect the incorporation of future earnings information in current prices, namely size (*Size*) as a proxy for a firm's information environment, measured as the logarithm of market value; earnings persistence (*Persistence*), which is an indicator variable that equals one if a company reports losses, and zero otherwise; earnings variability (*EarnVol*), measured as the standard deviation of earnings, given that greater earnings volatility makes it more difficult to predict earnings; firm's book to market ratio (*BM*); and *Opacity* as a proxy of financial transparency, measured based on scaled accruals, following Bhattacharya et al. (2003) and Dhaliwal et al. (2014).

To avoid our proxy of uncertainty avoidance capturing the effect of other country-level factors which could also influence the association between future earnings and current returns, we include the following variables as controls: *Disclosure* is a measure disclosure quality through a count of non-missing Compustat line items reflecting the extent of details in firms' financial statements (Chen et al., 2015; Li et al., 2019) and may affect investors ability to anticipate future earnings; Own is a proxy of ownership concentration measured by the proportion of closely held shares (La Porta et al., 1999; Djankov et al., 2005) and could affect earnings expectations (Dargenidou et al., 2007); NonCatholic is a proxy for religious decomposition measured as a dummy variable that takes one if a firm is from a country where the major religion is not Catholicism, and zero otherwise and could affect investors' economic attitudes correlated with uncertainty avoidance (Huang, 2008; Guiso et al., 2003); CivCom is a dummy variable that takes the value of zero if the country is characterized by civil law and one if common law; MarketDev is the market capitalization of listed companies as a percentage of GDP; InvProtection is a measure of legal protection of minority shareholders against expropriation by corporate insiders which is an index of the strength of minority shareholder protection against self-dealing by the controlling shareholder (La Porta et al., 2008). Enforcement is an index capturing the degree of the quality of audit function and degree of accounting enforcement in each country, developed by Brown et al. (2014). Motivated by prior literature arguing that behavioural biases and in particular ambiguity aversion could correlate with trading volume and volatility (Chui et al., 2010; Illeditsch; 2011; Antoniou et al., 2015), we follow Chui et al. (2010) and we include two more proxies: MarketTradingVolume, a proxy of market trading volume measured as the annual market dollar trading volume of Datastream Global index of a country divided by this index's market capitalization and; MarketVolatility, a proxy for market volatility measured as the sum of squared annual stock market returns of all firms a country divided by the number of firms. Finally, for the regression analyses, we

transform all continuous control variables in percentile ranks ranging from zero to one (see also Lundholm & Myers, 2002; Tucker & Zarowin, 2006; Haw et al., 2012).

To test our second hypothesis (H2), we extend Equation (2) and introduce *Openness*, our proxy of market openness, both as a main effect and as an interaction with  $E_{i,t+1,+3}$  and  $E_{i,t+1,+3}*UA$ , as follows:

$$R_{i,t}=a_0+b_1E_{i,t+1,+3}+b_2E_{i,t}+b_3E_{i,t-1}+b_4R_{i,t+1,+3}+b_5E_{i,t+1,+3}*UA+b_6E_{i,t+1,+3}*$$

$$Openness+b_7E_{i,t+1,+3}*UA*Openness+b_8UA+Openness+Controls+Industry FE \quad (3)$$

$$+Year FE+\varepsilon_{i,t}$$

where *Openness* is the measure for market openness and reflects the dominance of domestic investors relative to foreign commonly referred to as Equity Home Bias. Home bias measures investors' preference to be "reluctant to reap the full benefits of international diversification and overinvest in their domestic assets rather than in international portfolios" (Bose et al., 2015, p. 65). Thus, we employ the inverse of this measure: the dominance of domestic investors relative to foreign investors.<sup>7</sup> all other variables are as previously defined. For the regression analysis, we transform Equity Home Bias into reversed percentile ranks ranging from zero to one so that a higher percentile rank of captures a higher international portfolio diversification and hence openness. In Equation (3), the coefficient  $b_5$ , which is the coefficient of interaction between future earnings and uncertainty avoidance,  $E_{i,t+1,+3}*UA$ , captures the market's ability to anticipate future earnings in countries with low market openness and high uncertainty avoidance. The coefficient  $b_7$  is the coefficient of interaction between future earnings, uncertainty avoidance and market openness,  $E_{i,t+1,+3}*UA*Openness$ . This captures the incremental effect of market openness on the market's ability to anticipate

<sup>&</sup>lt;sup>7</sup> See page 6 in Kwabi et al. (2020) for a detailed discussion of how Equity Home Bias is calculated, using equity portfolio holding data obtained from the Coordinated Portfolio Investment Survey (CPIS) of the International Monetary Fund (IMF).

future earnings for firms from countries with higher uncertainty avoidance. In line with H2, which predicts that the effect of uncertainty avoidance would be moderated as the market becomes more open, we expect coefficient  $b_7$  to be positive and significant.<sup>8</sup>

Finally, all models include year and industry fixed effects in order to ensure that our relation is not driven by industry characteristics or by any particular sub-period during our sample period. We employ panel data analysis with OLS regressions and use robust standard errors clustered at the firm level across all estimations. Appendix A summarizes the definitions of all the variables used.

# 3.2 Sample selection process

Table 1 presents the sample selection process. The starting point for the sample is all companies listed in countries that adopted IFRS mandatorily from 2005 onwards. This was the first year that companies in the European Union and elsewhere started reporting under IFRS on a mandatory basis. More countries followed this mandatory transition to IFRS thereafter. We retrieve accounting and financial data from Datastream/Worldscope from the year of IFRS adoption until 2015. We then download data until 2018 to be able to calculate the sum of future earnings for the three years after the current year. Our sample includes all companies in the research lists of active and dead companies constructed by Datastream for each country in our sample.

-

<sup>&</sup>lt;sup>8</sup> We have no prediction for the coefficient  $b_6$ , which is the coefficient of interaction between future earnings and market openness,  $E_{i,t+1,+3}$ \* *Openness*.

<sup>&</sup>lt;sup>9</sup> Our conclusions remain unchanged when including firm fixed effects or clustering standard errors at firm and year level. Similarly, our inferences remain unchanged when using random effect or fixed effects models. The results are available upon request.

<sup>&</sup>lt;sup>10</sup> As in Mazzi et al. (2019) and Dionysiou et al. (2021), we assess whether a certain country has adopted IFRS mandatorily, we rely on the guide published by the IFRS Foundation on the use of IFRS by jurisdiction (http://www.ifrs.org/use-around-the-world/use-of-ifrs-standards-by-jurisdiction/)

From these lists, we eliminate instruments which are not classified as equity (i.e. DS item Type to be equal to 'EQ'). Further, we retain primary listings in the sample to avoid double counting firms that are cross-listed in more than one market. Subsequently, we exclude the first firm-year observation reported under IFRS to avoid confounding transitional effects. Following Daske et al. (2013) and Mazzi et al. (2019), we rely on Worldscope item 'accounting standards followed' (WC07536) to establish companies' financial reporting standards. Based on this item, we select the firm-year observations also used by Daske et al. (2013) and Mazzi et al. (2019) and eliminate those firm-year observations if this item is missing or returns a non-IFRS related code. This exclusion criterion ensures that firms' earnings are comparable across firms in our sample (Soderstrom & Sun, 2007). Subsequently, we exclude firms with missing industry classification and those belonging in the financials industry given that their accounting information is different from other firms and they are also subject to industry specific regulations which may affect investors' perceptions. Further, we exclude firm-year observations which changed their financial year end by eliminating those with accounting periods of more than 380 or less than 350 days (Dargenidou et al., 2018; Mazzi et al., 2019) compared to the current period, or any one of the subsequent three years following the current year period. This ensures that accounting earnings are reported for periods of similar length. In addition, we eliminate firm-year observations with missing firm-level data, country-level data and data with respect to the level of market openness. Finally, we winsorize all firm-level data at the 2% and 98% levels and following Haw et al. (2012), we eliminate firm-year observations when the absolute value of current or lagged earnings are above one and the absolute value of the sum of future earnings is above three. This ensures the omission of extreme values that may affect the findings. Our final sample comprises 26,882 firm-year observations corresponding to 4,879 firms across 20 countries.

#### TABLE 1 ABOUT HERE

### 4. Empirical results

# 4.1 Descriptive statistics

Table 2 shows descriptive statistics for earnings, stock returns and firm-specific control variables used in our models. In line with prior research (e.g., Haw et al., 2012; Dargenidou et al., 2021), we observe that earnings ( $E_{t+1,+3}$ ,  $E_t$  and  $E_{t-1}$ ) are left-skewed, given that their mean values are below median, in line with prior literature. Additionally, current and future returns ( $R_t$ ,  $R_{t+1,+3}$ ) are right-skewed, with mean values above the median, again consistent with prior literature. Further, approximately 26% of the firm-year observations report a loss, as indicated by the mean value of *Persistence*. The average firm-year observation in our sample has a book to market value of equity (BM) of 0.860.

### TABLE 2 ABOUT HERE

Table 3 presents the country-level descriptive statistics of the variables used in our analysis, along with the year of IFRS adoption and the number of observations for each country in our sample.

We find evidence that there is plenty of variation amongst our countries with respect to uncertainty avoidance (*UA*). The country with the lowest *UA* is Sweden, with an uncertainty avoidance score of 29, followed by Ireland and United Kingdom, where *UA* is 35. On the other side of the spectrum, the country with the highest value is Greece where *UA* is 112, followed by Portugal with *UA* score of 104. Further, we find that *Own*, our measure of ownership concentration, ranges from 57% to 66% for countries with higher uncertainty avoidance whilst for those with higher *UA* levels *Own* ranges from 19% to 35%. Thus, countries with higher ownership concentration tend to also have higher levels of uncertainty avoidance. Additionally, we find that values of *Enforcement* range from 11 to 45 for countries with higher uncertainty avoidance and from 34 to 54 for countries with lower levels of *UA*. This indicates that countries

with higher *UA* also exhibit weaker enforcement of accounting standards and lower quality of audit function.

Moreover, we observe no clear patterns with respect to *Openness* which ranges from 2.002 to 8.508 in countries with low uncertainty avoidance and 2.910 to 6.751 in countries with high uncertainty avoidance. Similarly, we do not observe clear patterns with respect to market trading volume, *MarketTradingVolume*, or market volatility, *MarketVolatility*, given the overlap of the values between countries with high and low uncertainty avoidance.

#### TABLE 3 ABOUT HERE

In order to further explore the patterns identified in Table 3, we split the sample across firms from countries with low and high uncertainty avoidance defining countries with low (high) uncertainty avoidance as those which exhibit a *UA* score below (above) the sample median. We present the descriptive statistics across the two subsamples in Table 4 (Panel A).

We find that companies operating in countries with high uncertainty avoidance earn lower returns than those in countries with lower levels of uncertainty avoidance (mean  $R_i$ : 0.108 vs 0.126, p-value of differences<0.05; mean  $R_{t+1,+3}$   $R_t$ : 0.358 vs 0.381, p-values of difference<0.01). Further, we find that companies operating in countries with high uncertainty avoidance relative to those with low uncertainty avoidance exhibit higher earnings volatility avoidance (mean EarnVol: 0.289 vs 0.239, p-value of differences<0.01), higher book to market ratios (mean BM: 1.031 vs 0.770; p-value of difference<0.01) and tend to be larger (mean Size: 12.925 vs 12.267; p-value of difference<0.01). Moreover, in line with the findings by Nabar and Boonlert-U-Thai (2007), we observe that the opacity of company earnings is significantly higher in countries with higher uncertainty avoidance as opposed to countries with uncertainty avoidance (mean Opacity: -0.028 vs -0.039; p-value of difference<0.01). The results also

-

<sup>&</sup>lt;sup>11</sup> By construction, a more positive (negative) value of opacity indicates greater (lower) earnings opacity.

suggest that trading volume is larger for countries with higher uncertainty avoidance than countries with lower uncertainty avoidance (mean *MarketTradingVolume*: 398.746 vs 90.859; p-value of difference<0.01) in line with prior literature indicating that behavioural biases generate excess trading volume (e.g. Chui et al., 2010). Additionally, market development, investor protection enforcement, are significantly lower in countries with higher uncertainty avoidance (mean *MarketDev* 0.048 vs 0.077; mean *InvProtection* 0.427 vs 0.558; mean *Enforcement*: 30.083 vs 45.364; all p-values of differences<0.01). These results show that uncertainty avoidance tends to be more prevalent in countries with weaker investor protection, weak enforcement and poor earnings quality. In line with the underlying premise in our hypothesis, these results indicate a financial reporting environment that would impair investors' ability to better anticipate future earnings in countries with high uncertainty avoidance.

#### **TABLE 4 ABOUT HERE**

In Table 4 (Panel A), we find that market openness is lower for firms in countries with high uncertainty avoidance (mean *Openness*: 4.849 vs 3.207; p-value of difference<0.01).<sup>12</sup> Considering this and our second hypothesis, Table 4 (Panel B) reports the descriptive of the variables used in our analysis across firms from countries with high and low market openness. We define countries with low (high) market openness as those which exhibit an *Openness* score below (above) the sample median across all countries in our sample at any given year.

We observe that the level of uncertainty avoidance is significantly lower for countries with greater market openness (mean *UA*: 57.472 vs 75.919, p-value of difference<0.01). Consistently, we find that countries with greater market openness exhibit lower levels of ownership concentration (mean *Own*: 45.126 vs 59.399; p-values of difference<0.01), higher

<sup>&</sup>lt;sup>12</sup> Openness is measured using Equity Home Bias and therefore a larger (smaller) value is indicative of greater (lower) proportion of home investors and thus lower (greater) market openness.

market development (mean *MarketDev*: 0.075 vs 0.049; p-values of difference<0.01), higher investor protection (mean *InvProtection*: 0.544 vs 0.443; p-values of difference<0.01), and stronger enforcement infrastructures (mean *Enforcement*: 44.226 vs 30.996; p-values of difference<0.01). Further, we observe that firms from countries with greater market openness exhibit lower earnings opacity (mean *Opacity*: 0.038 vs -0.030; p-value of difference<0.01). Taken together, such evidence indicates that investment in these firms may be less risky, which is consistent with these firms generating lower returns (mean  $R_i$ : 0.107 vs 0.147; p-value of difference<0.01).

# 4.2 The relation between uncertainty avoidance and price informativeness

Table 5 shows the results of the empirical implementation of Equation (1) and Equation (2) which tests the effect of uncertainty avoidance on the share price anticipation of future earnings. The results presented in the first column yield correlations in line with the findings in prior literature (e.g., Lundholm & Myers, 2002; Ettredge et al., 2005; Haw et al., 2012; Dargenidou et al., 2021). Specifically, the coefficient of future earnings,  $E_{i,t+1,+3}$ , is positive and significant reflecting the favourable market's response to information about future earnings that is anticipated and not reflected in/captured by current and past earnings (coefficient: 0.169; p-values<0.01). The coefficient of  $E_{i,t}$  is positive and significant capturing the positive market response to the unexpected portion of current earnings (coefficient: 0.598; p-values<0.01). Further the coefficient of  $E_{i,t-1}$  captures the already anticipated portion of current earnings and is negative and significant is as expected (coefficient: -0.417; p-values<0.01). Finally, the

\_

<sup>&</sup>lt;sup>13</sup> Appendix B reports a correlations matrix between the variables employed in our analysis. We do not note extreme correlations between the majority of the variables. The only exceptions are the high correlations (perhaps not so surprisingly) between CivCom and InvProtection (-0.954\*\*\*) and between Openness and Enforcement (-0.769\*\*\*). To alleviate any concerns that the results presented below may be affected by these correlations, we have repeated our analysis by excluding CivCom and Enforcement. Our findings remain the same after these exclusions. Perhaps this is not surprising given that both variables are time-invariant and our conclusions hold when using fixed effects models or firm fixed effects (see footnote 9).

coefficient of  $R_{i,t+1,+3}$  is negative and significant as expected showing that realized future earnings contain a measurement error that future returns remove (coefficient: -0.033; p-values<0.01).

The results presented in column (2) confirm our hypothesis H1. Specifically, as expected, we find that the coefficient of the interaction between uncertainty avoidance and future earnings,  $UA * E_{t+1,+3}$ , is negative and significant (coefficient: -0.069; p-values<0.01), while the coefficient of future earnings,  $E_{t+1,+3}$ , is positive and significant (coefficient: 0.203; p-values<0.01). These results indicate that the market's anticipation of future earnings is weaker for firms in countries with higher uncertainty avoidance and, thus, such firms experience relatively lower stock price informativeness about future earnings. 14

### TABLE 5 ABOUT HERE

The remaining columns in Table 5 present the effect of uncertainty avoidance on the share price anticipation of future earnings when we include firm and country controls. Column 3 presents the results when we add only firm controls, and Column four presents the results after controlling for both firm and country characteristics. In line with the results presented in Column 2, the coefficient of future earnings is positive and significant (0.260, p-value<0.01; 0.301; p-value<0.01,), as expected. Importantly, we find that the coefficient of interaction between uncertainty and future earnings is negative and significant (-0.081, p-value<0.01; -0.103, p-value<0.01, in the third and fourth columns respectively). To demonstrate the economic significance of this finding, a one standard deviation increase in this interaction variable induces a 2.4% decrease in current returns (based on the coefficient in Column 3; this becomes 3% if we used the coefficient in Column 4).<sup>15</sup>

<sup>&</sup>lt;sup>14</sup> We have also examined the potential of non-linear relationship between returns and our focal variable of interest i.e. the interaction between future earnings and uncertainty avoidance by introducing its square term in Model (2) of Table 5. Our results indicate that non-linearity is not present.

<sup>&</sup>lt;sup>15</sup> Drawing upon the descriptive statistics presented in Table 2, the decrease of 2.4% is calculated as -0.081\*0. 296 = -0.024. The decrease of 3% is calculated as -0.103\*0.296 = -0.030.

Regarding the remaining variables in the third and fourth columns respectively, these present coefficients that are well in line with previously literature (e.g. Lundholm & Myers, 2002; Ettredge et al., 2005; Dargenidou et al., 2021). More specifically, the coefficients of current earnings remain significantly positive at the 1% level, while the coefficients of lagged earnings also remain significantly negative at 1% across all estimations. Lastly, the coefficients of future returns remain significantly negative at the 1% level, showing that realized future earnings contain a measurement error that future returns remove.

Further, in line with Ettredge et al. (2005), we find that the relation between current returns and future earnings to be stronger for larger firms which exhibit richer information environments (coefficients: 0.088 and 0.075; p-values<0.01). Further, we find that the association between current returns and future earnings is weaker for firms that report losses (coefficients: -0.089 and -0.091; p-values<0.01). This is in line with prior literature given that predicting future losses is more difficult compared to predicting future profits (e.g. Ettredge et al., 2005; Lundholm & Myers. 2002). Additionally, firms with higher book to market exhibit a weaker association between current returns and future earnings (coefficients: -0.041 and -0.066; p-values<0.10 and p-values<0.01, respectively). Taken together, these results indicate that the market's anticipation of future earnings is weaker for firms from countries with high uncertainty avoidance and, thus, such firms experience relatively low stock price informativeness about future earnings, even after controlling for firm and country characteristics that may affect the anticipation of future earnings.

#### TABLE 6 ABOUT HERE

Table 6 reports the results when we also control for market openness. The first column shows the results when we include market openness as a control variable. The results show that the coefficient of interaction between future earnings and uncertainty avoidance,  $UA * E_{t+1,+3}$ , remains negative and significant (coefficient: -0.103; p-value<0.01). Further, our results show

that the coefficient of interaction between future earnings and market openness, *Openness* \*  $E_{t+1,+3}$ , is insignificant (coefficient: -0.058; p-value>0.10), indicating that market openness does not improve the market's ability to anticipate future earnings.

Column 2 of Table 6 reports the results of estimation of Equation (3), which examines whether market openness weakens the effect of uncertainty avoidance on the share price anticipation of future earnings. Our results show that the coefficient of the interaction between uncertainty avoidance, future earnings and market openness, Openness \*  $E_{t+1,+3}*UA$ , is positive and significant (coefficient: 0.255, p-value<0.05). Further, the coefficient of interaction between uncertainty avoidance and future earnings,  $UA * E_{t+1,+3}$ , is negative and significant (coefficient: -0.206, p-value<0.01), while the coefficient of future earnings,  $E_{t+1,+3}$ , is positive and significant (coefficient: 0.379, p-value<0.01). Taken together, these findings are in line with our second hypothesis and indicate that the negative effect of uncertainty avoidance is more pronounced within markets with lower participation from foreign investors, and less pronounced within highly open markets. Thus, our results are consistent with the view that the effect of national culture is reduced when a country's market is more open. Importantly, it corroborates our first hypothesis in that it is indeed investors' cultural attributes explaining the weaker relation between current returns and future earnings in countries with stronger uncertainty avoidance as opposed to managers' cultural attributes inducing a lower quality of information environment which then results in weaker relation between current returns and future earnings.

Consistent with the descriptive statistics presented in Table 4, the parameter coefficient of the interaction variable between market openness and uncertainty is significantly negative (coefficient: -0.251, p-value<0.01). This indicates that countries with lower uncertainty avoidance exhibit a higher presence of foreign investors.

#### 4.3 Robustness tests

We undertake a number of robustness tests to examine the sensitivity of our findings. We present the corresponding results in Tables 7, 8 and 9. First, our tests focus on one aspect of culture and, arguably, may be driven by ignoring the effects of other cultural aspects which may drive the relation between current returns and future earnings. Thus, we expand the empirical model by also considering additional cultural variables, namely individualism and power distance and present the effect of uncertainty avoidance on the share price anticipation of future earnings when controlling for additional cultural variables in Table 7. Our conclusions remain unchanged: the parameter coefficient of the interaction variable between uncertainty avoidance and future earnings remains significantly negative (coefficients: -0.133 and -0.294; p-values<0.01), while the effect of market openness upon the relation between future earnings and returns remain positive and significant (coefficients: 0.377; p-value<0.01). Thus, our results are unlikely to be driven by other omitted cultural factors.

#### TABLE 7 ABOUT HERE

Columns 1 and 2 of Table 8, repeats our estimation when excluding countries with fewer than 150 observations, and re-estimate the results to test the counterpart scenario of countries with very few observations that may drive our main results. Our conclusions remain unchanged under this exclusion: the parameter coefficient of the interaction variable between uncertainty avoidance and future earnings remains significantly negative (coefficients: -0.098 and -0.377; p-values<0.01), while the effect of market openness upon the relation between future earnings and returns remain positive and significant (coefficients: 0.246; p-value<0.05). Further, in our main tests, we measure uncertainty avoidance by employing the relevant index from Globe's cultural dimensions. To examine the sensitivity of our results, we employ Hoftsede's uncertainty avoidance cultural dimension despite being subject to criticism (see McSweeney, 2002; Baskerville, 2003; Mazzi et al., 2018 for discussion). Columns 3 and 4 report the

corresponding results. We find that our conclusions hold when using Hofstede (1980) uncertainty avoidance index. Specifically, the coefficient of interaction between uncertainty avoidance and future earnings remains significantly negative (coefficients: -0.104 and -0.202; p-values<0.01), while the effect of market openness upon the relation between future earnings and returns remain positive and significant (coefficients: 0.235; p-value<0.01). Additionally, we undertake robustness tests regarding the market openness proxy used earlier. In line with El Ghoul and Zheng (2016) we use exports of goods and services to GDP, as an alternative proxy for market openness and find similar results. Specifically, the coefficient of interaction between future earnings, uncertainty avoidance and market openness remain positive and significant in column 5 (coefficients: 0.240; p-value<0.10).

### TABLE 8 ABOUT HERE

Table 9 presents the effect of uncertainty avoidance and the moderating role of market openness at the aggregate country level. The dependent variable is the future earnings response coefficient estimated using Equation (1) for each country-year portfolio requiring a minimum of 20 observations. The independent variables include all country-level variables used in our earlier analysis. Column 1 shows that coefficient of uncertainty avoidance is albeit insignificant negative (coefficient: -0.017; p-value>0.10). However, we find that the adverse effect of uncertainty avoidance becomes significant when we control for market openness. Specifically, Column 2 shows that the coefficient of the interaction between uncertainty avoidance and market openness, *UA* \**UA*, is positive and significant (coefficient: 0.486; p-value<0.05), while the coefficient of uncertainty avoidance is negative and significant (coefficient: -0.229; p-value<0.05). Further, our conclusions remain unchanged when we repeat our analysis by excluding the UK. These results indicate that the market's anticipation of future earnings is weaker in countries with high uncertainty avoidance, and that this relationship is weaker in

countries with higher market openness. Overall, these results are consistent with the analysis presented earlier using firm-level data as opposed to aggregate country-level data.

#### TABLE 9 ABOUT HERE

#### 5. Conclusion

In this study, we explore the effect of a distinct feature of national culture and uncertainty avoidance in particular upon the market ability to anticipate future earnings as reflected in current returns. We find that uncertainty avoidance does indeed affect the stock price informativeness about future earnings future earnings. Specifically, we report that current stock returns are more strongly related with future earnings in countries with low uncertainty avoidance and weaker for countries with higher levels of uncertainty avoidance. Markets with investors embedded in cultures with lower uncertainty avoidance are thus relatively more informationally efficient. We find that our relation is less pronounced in more open markets which are associated with relatively high levels of foreign stock market participation, offering further evidence that the effect of culture is reduced when a market is more open.

Our results are of interest to researchers working in the fields of finance, accounting and cultural studies. If more information about fundamentals and a company's future prospects is reflected in current stock returns, prices exhibit higher informativeness and informational efficiency, which facilitates more efficient resource allocation (Durnev et al., 2003). The informational efficiency of stock markets is a major research area. In this study, we show evidence that culture influences investors' anticipation of future earnings how this is reflected in returns. Most studies in finance tend to report that culture is related with the magnitude of abnormal stock returns of investment strategies. Our study instead explores the link between one distinct feature of culture and the informational efficiency about future earnings of stock prices. We show that national culture is a determinant of share price informativeness about

future earnings which has not been considered thus far by prior literature (e.g., Ettredge et al., 2005; Choi et al., 2011; Haw et al., 2012; Choi et al., 2013). Future research could expand our knowledge on the effect of investors' innate characteristics on their investment behaviour by applying different research methods such as a quantile-on-quantile based approach as in Atsalakis et al. (2020).

Definition Source									
Dependent variable									
$R_t^I$	One-year compounded firm stock returns, measured from three months after the end of each firm's financial year end onwards.	Return index (RI): Datastream							
Main independent varia	bles								
UA	Percentile rank of uncertainty avoidance.	Uncertainty avoidance index reported in House et al. (2004) Income before extraordinary items:							
$E_t^{I}$	The net income scaled by firm market value.	WC01551 Market Capitalization: WC08001							
$E_{i,t+I,+3}$	The sum of future earnings of firm <i>i</i> for the 3 years following the current year, scaled by the market value of equity measured at the firm's year end.	Income before extraordinary items: WC01551 Market Capitalization: WC08001							
$R_{i,t+1,+3}$	The compounded firm stock return measured for the 3 years following the current year.	Return index (RI): Datastream							
Control variables									
Size	Percentile rank of the natural logarithm of market value of equity.	Market Capitalization: WC08001							
Persistence	A dummy variable that takes 1 if a firm report losses and 0 otherwise.	Income before extraordinary items: WC01551							
EarnVol <sup>1</sup>	Percentile rank of the standard deviation of net income scaled by firm market value.	Income before extraordinary items: WC01551 Market Capitalization: WC08001							
$BM^{1}$	Percentile rank of the book to market ratio.	Common equity: WC03501 Market capitalisation WC08001							
$Opacity^{I}$	Percentile rank of firm-level earnings opacity. Following Bhattacharya Daouk and Welker (2003) and Dhaliwal et al. (2014), we measure earnings opacity as follows: ( $\Delta$ CA- $\Delta$ CL- $\Delta$ CASH+ $\Delta$ STD-DEP+ $\Delta$ TP)/lag(TA), where $\Delta$ CAis the change in total current assets; $\Delta$ CLis the change in total current liabilities; $\Delta$ CASH is the change in cash; $\Delta$ STD is the change in the current portion of long-term debt included in total current liabilities; DEP is the depreciation and amortization expense; $\Delta$ TPis the change in income taxes payable; and lag(TA) is the total assets at the end of the previous year.	Current assets: WC02201 Current liabilities: WC03101 Cash and short term investments: WC02001 Short term debt: WC03051 Depreciation and amortisation: WC01151 Tax payable: WC03063 Total assets: WC02999							
Disclosure	Percentile rank of the yearly median value of disclosure quality measured for each country separately. We measure disclosure quality following Chen S. et al. (2015) as adapted for international firms by Li et al. (2019) by counting non-missing Compustat line item.	Constructed using Compustat Global							
Own	Percentile rank of the yearly median value of closely held shares measured for each country separately.	Closely held shares: WC08021							
MarketTradingVolume	Percentile rank of market trading volume. Following Chui et al. (2010), we measure market trading volume as the annual market dollar trading volume of the Datastream Global index of the country divided by this index's market capitalisation.	Trading volume of the Datastream Global index of the country: VO Market capitalisation of the Datastream Global index of the country: MV							
MarketVolatility	Percentile rank of market trading volatility. Following Chui et al. (2010), we measure market trading volatility for each country and year as the sum of squared stock returns of all firms in the	Return index (RI): Datastream							
NonCatholic	country divided by the number of firms list in that country. A dummy variable that takes the value of 1 if the major religion is Catholicism and zero otherwise.	Central Intelligence Agency website							
		20							

CivCom	A dummy variable that takes the value of 0 if civil law and 1 if common law.	La Porta et al. (1998)
MarketDev	Percentile rank of the ratio of market capitalisation of Datastream Global index of a country to GDP.	Market capitalisation of Datastream Global index of a country: MV GDP: World Bank
InvProtection	Percentile rank of a measure of legal protection of minority shareholders against expropriation by corporate insiders.	Anti self-dealing index: La Porta et al. (2008)
Enforcement	Percentile rank of the degree of the quality of audit function and degree of accounting enforcement in each country.	Brown et al. (2014)
Openness	Reversed percentile rank of home bias. Following Kwabi et al. (2020), we measure home bias as the degree to which domestic investors hold a significantly higher percentage of domestic securities in their portfolios relative to the theoretical prescription of the International Capital Asset Pricing Model (ICAPM) benchmark.	Coordinated Portfolio Investment Survey (CPIS) of the International Monetary Fund (IMF)

<sup>&</sup>lt;sup>1</sup> Variables winsorized at the 2% and 98% levels.

**Appendix B: Correlation matrix** 

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
$E_{t+1,+3}\left(1\right)$	1									
$E_t(2)$	0.466***	1								
$E_{t-1}(3)$	0.351***	$0.474^{***}$	1							
$R_t(4)4$	$0.176^{***}$	$0.174^{***}$	-0.0313***	1						
$R_{i,t+1,+3}\left(5\right)$	0.424***	0.116***	$0.120^{***}$	-0.0341***	1					
Persistence (6)	-0.407***	-0.719***	-0.403***	-0.153***	-0.111***	1				
EarnVol (7)	-0.354***	-0.193***	-0.162***	-0.0647***	-0.182***	$0.208^{***}$	1			
BM (8)	-0.103***	-0.0990***	0.0189***	-0.191***	0.137***	0.121***	0.0996***	1		
Size (9)	0.221***	0.254***	0.238***	$0.107^{***}$	-0.000332	-0.321***	-0.193***	-0.297***	1	
Opacity (10)	0.0327***	0.171***	0.0747***	-0.0284***	-0.0324***	-0.142***	0.00979	$0.0119^*$	$0.0149^{**}$	1
UA (11)	-0.0179***	-0.00232	0.00527	-0.0280***	-0.0516***	-0.0160***	0.0384***	$0.170^{***}$	-0.0851***	0.0232***
Disclosure (12)	-0.0283***	-0.0205***	-0.0252***	-0.0206***	-0.0487***	$0.0105^{*}$	0.0259***	-0.0165***	-0.0731***	$0.0119^{*}$
Own (13)	0.0526***	0.0266***	0.0183***	0.0354***	0.0320***	-0.0217***	-0.00874	0.138***	-0.0310***	0.0308***
MarketTradingVolume (14)	0.0198***	-0.00286	0.000369	0.0457***	0.00402	0.00589	-0.0143**	0.0170***	0.0111*	0.00853
MarketVolatility (15)	0.0197***	0.0193***	-0.00283	0.0944***	$0.0128^{**}$	0.00266	-0.00282	-0.00496	0.0831***	0.00876
NonCatholic (16)	-0.0577***	-0.0496***	-0.0401***	$0.0209^{***}$	0.00184	0.0930***	$0.0480^{***}$	0.0243***	-0.139***	0.0264***
CivCom (17)	$0.0152^{**}$	0.00626	0.0121**	-0.0257***	-0.0106*	-0.0461***	-0.0267***	0.0382***	0.122***	-0.00707
MarketDev (18)	-0.0210***	-0.00319	-0.0329***	$0.0862^{***}$	-0.0943***	0.00706	0.0181***	-0.175***	0.00304	-0.0110°
InvProtection (19)	-0.0173***	-0.0149**	-0.0222***	0.0197***	0.0135**	0.0536***	0.00198	-0.0590***	-0.146***	0.00187
Enforcement (20)	-0.0285***	-0.0282***	-0.0440***	-0.0169***	-0.0314***	0.0287***	-0.0175***	-0.106***	-0.196***	-0.0519**
Openness (21)	0.0189***	0.00131	0.00916	0.0205***	0.0347***	0.00683	0.0159***	0.167***	0.0915***	0.0383***

continued

	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)
UA (11)	1										
Disclosure (12)	0.0399***	1									
Own (13)	0.712***	0.0961***	1								
MarketTradingVolume (14)	0.0474***	0.0595***	0.112***	1							
MarketVolatility (15)	-0.0638***	-0.119***	0.136***	0.0349***	1						
NonCatholic (16)	-0.382***	$0.0708^{***}$	-0.358***	0.0537***	-0.0312***	1					
CivCom (17)	0.545***	-0.0443***	0.334***	0.0311***	-0.0129**	-0.619***	1				
MarketDev (18)	-0.697***	-0.0590***	-0.617***	-0.106***	-0.00351	0.473***	-0.667***	1			
InvProtection (19)	-0.575***	-0.0000722	-0.397***	-0.0120**	-0.0309***	0.626***	-0.954***	0.702***	1		
Enforcement (20)	-0.415***	0.0512***	-0.300***	-0.116***	-0.0658***	-0.0116*	-0.538***	0.489***	0.573***	1	
Openness (21)	0.506***	0.0588***	0.550***	0.138***	0.130***	0.0810***	0.386***	-0.593***	-0.477***	-0.769***	1

#### References

- Abdallah, A. A. N., Abdallah, W., & Saad, M., 2021. The impact of national culture on the synchronicity of cross-listed firms. *Finance Res. Lett.*, https://doi.org/10.1016/j.frl.2021.102293.
- Abdolmohammadi, M.J. and Sarens, G., 2011. An investigation of the association between cultural dimensions and variations in perceived use of and compliance with internal auditing standards in 19 countries. *The International J. of Account.* 46(4), 365-389.
- Aggarwal, R., Faccio, M., Guedhami, O., & Kwok, C. C., 2016. Culture and finance: An introduction. *J. of Corporate Finance* 100(41), 466-474.
- Anderson, C. W., Fedenia, M., Hirschey, M., & Skiba, H., 2011. Cultural influences on home bias and international diversification by institutional investors. *J. of Bank. & Finance* 35(4), 916-934.
- Antoniou, C., Harris, R.D. and Zhang, R., 2015. Ambiguity aversion and stock market participation: An empirical analysis. *J. of Bank. & Finance* 58, 57-70.
- Atsalakis, G. S., Bouri, E., & Pasiouras, F., 2020. Natural disasters and economic growth: a quantile on quantile approach. *Annals of Operations Res.*, 1-27.
- Ball, R., 2006. International Financial Reporting Standards (IFRS): Pros and cons for investors. *Account. and Bus. Res.* 36(sup1), 5-27.
- Baskerville, R.F., 2003. Hofstede never studied culture. *Account.*, *Organizations and Society* 28(1), 1-14.
- Beckmann, D., Menkhoff, L. and Suto, M., 2008. Does culture influence asset managers' views and behavior?. *Journal of Economic Behavior & Organization* 67(3-4), 624-643.
- Beugelsdijk, S. and Frijns, B., 2010. A cultural explanation of the foreign bias in international asset allocation. *J. of Bank. & Finance* 34(9), 2121-2131.
- Bhattacharya, U., Daouk, H. and Welker, M., 2003. The world price of earnings opacity. *Account. Rev. The Account. Rev.* 78(3), 641-678.
- Bose, U., MacDonald, R., & Tsoukas, S., 2015. Education and the local equity bias around the world. *J. of International Financial Mark.*, *Inst. and Money*, *39*, 65-88.
- Brochet, F., Jagolinzer, A. D., & Riedl, E. J., 2013. Mandatory IFRS adoption and financial statement comparability. *Contemporary Account. Res.*, 30(4), 1373-1400.
- Brown, P., Preiato, J. and Tarca, A., 2014. Measuring country differences in enforcement of accounting standards: An audit and enforcement proxy. *J. of Bus. Finance & Account.* 41(1-2), 1-52.
- Bushman, R., Piotroski, J. and Smith, A., 2004. What determines corporate transparency? *J. of Account. Res.* 42(2), 207-252.
- Chang, K. and Noorbakhsh, A., 2009. Does national culture affect international corporate cash holdings? *J. of Multinatl. Financial Management* 19(5), 323-342.
- Chen, S., Miao, B.I.N. and Shevlin, T., 2015. A new measure of disclosure quality: The level of disaggregation of accounting data in annual reports. *J. of Account. Res.* 53(5), 1017-1054.
- Chen, Y., Dou, P.Y., Rhee, S.G., Truong, C. and Veeraraghavan, M., 2015. National culture and corporate cash holdings around the world. *J. of Bank. & Finance 50*, 1-18.
- Cheon, Y-H. and Lee, K-H., 2018. Maxing out globally: Individualism, investor attention, and the cross section of expected stock returns. *Management Science* 64(12), 5807-5831.
- Choi, J. H., Choi, S., Myers, L. A., & Ziebart, D., 2019. Financial statement comparability and the informativeness of stock prices about future earnings. *Contemporary Account. Res.*, 36(1), 389-417.

- Choi, J., Myers, L., Zhang, Y. and Ziebert, D., 2011. Do management EPS forecasts allow returns to reflect future earnings? Implications for the continuation of management's quarterly earnings guidance. *Rev. of Account. Stud.* 16(1), 143-182.
- Choi, Y.S., Peasnell, K. and Toniato, J., 2013. Has the IASB been successful in making accounting earnings more useful for prediction and valuation? UK evidence. *J. of Bus. Finance & Account.* 40(7-8), 741-768.
- Chou, T.K., 2013. Information content of credit ratings in pricing of future earnings. *Rev. of Quantitative Finance and Account.* 40(2), 217-250.
- Chui, A., Titman, S. and Wei, J., 2010. Individualism and momentum around the world. *J. of Finance* 65(1), 361-392.
- Collins, D., Kothari, S., Shanken, J. and Sloan, R., 1994. The lack of timeliness and noise as explanations for the low contemporaneous returns-earnings association. *J. of Account. and Economics* 18(3), 289-324.
- Dionysiou, D., Slack, R., Tsalavoutas, Y., & Tsoligkas, F. (2021). The capitalisation of intangibles debate: software development costs. ACCA and Adam Smith Business School. (doi:10.36399/gla.pubs.244180).
- Cowen, T., 2009. *Creative destruction: How globalization is changing the world's cultures*. Princeton University Press.
- Dargenidou, C., Jackson, R. H., Tsalavoutas, I., & Tsoligkas, F. (2021). Capitalisation of R&D and the informativeness of stock prices: pre-and post-IFRS evidence. *The Br. Account. Rev.*, https://doi.org/10.1016/j.bar.2021.100998.
- Dargenidou, C., McLeay, S. and Raonic, I., 2007. Ownership, investor protection and earnings expectations. *J. of Bus. Finance & Account.* 34(1-2), 247-268.
- Dargenidou, C., Tonks, I. and Tsoligkas, F., 2018. Insider trading and the post-earnings announcement drift. *J. of Bus. Finance & Account.* 45(3-4), 482-508.
- Daske, H., Hail, L., Leuz, C. and Verdi, R., 2013. Adopting a label: Heterogeneity in the economic consequences around IAS/IFRS adoptions. *J. of Account. Res.* 51(3), pp.495-547.
- Dhaliwal, D., Li, O.Z., Tsang, A. and Yang, Y.G., 2014. Corporate social responsibility disclosure and the cost of equity capital: The roles of stakeholder orientation and financial transparency. *J. of Account. and Public Policy* 33(4), 328-355.
- Djankov, S., La Porta, R., Lopez-de-Silanes, F. and Shleifer, A., 2008. The law and economics of self-dealing. *J. of Financial Economics* 88(3), 430-465.
- Dou, P., Truong, C. and Veeraraghavan, M., 2016. Individualism, uncertainty avoidance, and earnings momentum in international markets. *Contemporary Account. Res.* 33(2) 851-881.
- Durney, A., Morck, R., Yeung, B. and Zarowin, P., 2003. Does greater firm-specific return variation mean more or less informed stock pricing?. *J. of Account. Res.*, 41(5), 797-836.
- El Ghoul, S. and Zheng, X., 2016. Trade credit provision and national culture. *J. of Corporate Finance* 41, 475-501.
- Epstein, L. G. and Schneider, M. (2008). Ambiguity, information quality, and asset pricing, *The J. of Finance*, 63(1), pp. 197-228.
- Ettredge, M., Kwon, S., Smith, D. and Zarowin, P., 2005. The impact of SFAS No 131 business segment data on the market's ability to anticipate future earnings. *Account. Rev.* 80(3), 773-804.
- Eun, C., Wang, L. and Xiao, S., 2015. Culture and R<sup>2</sup>. *J. of Financial Economics* 115(2), 283-303.
- Frijns, B., Gilbert, A., Lehnert, T. and Tourani-Rad, A., 2013. Uncertainty avoidance, risk tolerance and corporate takeover decisions. *J. of Bank. & Finance* 37(7), 2457-2471.

- Gaganis, C., Leledakis, G. N., Pasiouras, F., & Pyrgiotakis, E. G., 2021. National culture of secrecy and stock price synchronicity: Cross-country evidence. *Working paper*
- Gassen, J., Skaife, H.A. and Veenman, D., 2020. Illiquidity and the measurement of stock price synchronicity. *Contemporary Account. Rev.* 37(1),419-456.
- Gelb, D.S. and Zarowin, P., 2002. Corporate disclosure policy and the informativeness of stock prices. *Rev. of Account. Stud.* 7(1), 33-52.
- Guiso, L., Sapienza, P. and Zingales, L., 2003. People's opium? Religion and economic attitudes. *J. of Monetary Economics* 50(1), 225-282.
- Harrison, G.L. and McKinnon, J.L., 1999. Cross-cultural research in management control systems design: A review of the current state. *Account., Organizations and Society* 24(5-6), 483-506.
- Harrison, G.L., McKinnon, J.L., Panchapakesan, S. and Leung, M., 1994. The influence of culture on organizational design and planning and control in Australia and the United States compared with Singapore and Hong Kong. *J. of International Financial Management & Account.* 5(3), 242-261.
- Haw, I., Hu, B., Lee, J. and Wu, W., 2012. Investor protection and price informativeness about future earnings: International evidence. *Rev. of Account. Stud.* 17(2), 389-419.
- Hofstede, G., 1980. Culture's consequences: International differences in work related values, Sage: Beverley Hills, CA.
- Hofstede, G., 1984. Cultural dimensions in management and planning. *Asia Pac. J. of Management* 1(2), 81-99.
- Hope, O., 2003. Disclosure practices, enforcement of accounting standards, and analysts' forecast accuracy: An international study. *J. of Account. Res.* 41(2), 235-272.
- House, R. J., Hanges, P. J., Javidan, M., Dorfman, P. W., & Gupta, V. 2004., Culture, leadership, and organizations: The GLOBE study of 62 societies. Thousand Oaks, CA: Sage Publications
- Huang, R.R., 2008. Tolerance for uncertainty and the growth of informationally opaque industries. *J. of Development Economics* 87(2), 333-353.
- Hussainey, K. and Walker, M., 2009. The effects of voluntary disclosure and dividend propensity on prices leading earnings. *Account. and Bus. Res.* 39(1), 37-55.
- Illeditsch, P.K., 2011. Ambiguous information, portfolio inertia, and excess volatility. *The J. of Finance* 66(6), 2213-2247.
- Jones, E.L., 2009. *Cultures merging: A historical and economic critique of culture* (Vol. 26). Princeton University Press.
- Kanagaretnam, K., Lim, C. Y., & Lobo, G. J., 2014. Influence of national culture on accounting conservatism and risk-taking in the banking industry. *Account. Rev. The Account. Rev.*, 89(3), 1115-1149.
- Kim, I., Pantzalis, C. and Wang, B., 2018. Shareholder coordination and stock price informativeness. *J. of Bus. Finance & Account.* 45(5-6), 686-713.
- Kogut, B. and Singh, H., 1988. The effect of national culture on the choice of entry mode. *J. of International Bus. Stud.* 19(3), 411-432.
- Kothari, S.P., Lewellen, J. and Warner, J.B., 2006. Stock returns, aggregate earnings surprises, and behavioral finance. *J. of Financial Economics* 79(3), 537-568.
- Kwabi, F.O., Thapa, C., Paudyal, K. and Neupane, S., 2020. Suboptimal international equity portfolio diversification and stock market development. *Rev. of Quantitative Finance and Account.* 54(1), 389-412
- La Porta, R., Lopez-de-Silanes, F. and Shleifer, A., 1999. Corporate ownership around the world. *The J. of Finance* 54(2), 471-517.

- La Porta, R., Lopez-De-Silanes, F., & Shleifer, A., 2008. The economic consequences of legal origins. *J. of Economic Literature* 46(2), 285-332.
- Li, B., Siciliano, G. and Venkatachalam, M., 2019. Economic Consequences of IFRS Adoption: The Role of Changes in Disclosure Quality. *Available at SSRN 3366288*.
- Lundholm, R. and Myers, L.A., 2002. Bringing the future forward: The effect of disclosure on the returns-earnings relation. *J. of Account. Res.* 40(4), 809-839.
- Maher, D., & Parikh, A., 2011. Short-term under/overreaction, anticipation or uncertainty avoidance? Evidence from India. *J. of International Financial Mark.*, *Inst. and Money*, 21(4), 560-584.
- Makrychoriti, P., & Pasiouras, F., 2021. National culture and central bank transparency: Crosscountry evidence. *J. of International Financial Mark.*, *Inst. and Money*, 72, in press.
- Mazzi, F., Slack, R. and Tsalavoutas, I., 2018. The effect of corruption and culture on mandatory disclosure compliance levels: Goodwill reporting in Europe. *J. of International Account.*, *Auditing and Tax.* 31, 52-73.
- Mazzi, F., Slack, R., Tsalavoutas, I. and Tsoligkas, F., 2019. Country-level corruption and accounting choice: Research & development capitalization under IFRS. *The Br. Account. Rev.* 51(5), 1-25.
- McSweeney, B., 2002. Hofstede's model of national cultural differences and their consequences: A triumph of faith-a failure of analysis. *Human relations* 55(1), 89-118.
- Moreira, J. C. C., Lima, G. A., & Góis, A. D., 2019. Effects of institutional factors on the accruals anomaly in Latin America. *J. of International Account., Auditing and Tax.*, 36, 100269.
- Nabar, S., & Boonlert-U-Thai, K. K. (2007). Earnings management, investor protection, and national culture. *J. of International Account. Res.*, 6(2), 35-54.
- Neel, M. (2017). Accounting comparability and economic outcomes of mandatory IFRS adoption. *Contemporary Account. Res.*, *34*(1), 658-690.
- Nguyen, N.H. and Truong, C., 2013. The information content of stock markets around the world: A cultural explanation. *J. of International Financial Mark.*, *Inst. and Money* 26, 1-29
- Nobes, C., 2006. The survival of international differences under IFRS: Towards a research agenda. *Account. and Bus. Res.* 36(3), 233-245.
- Papanastasopoulos, G. A., 2014. Accounting accruals and stock returns: Evidence from European equity markets. *European Account. Rev.*, 23(4), 729-768.
- Pasiouras, F., Bouri, E., Roubaud, D., & Galariotis, E., 2020. Culture and Multiple Firm—Bank Relationships: A Matter of Secrecy and Trust?. *J. of Bus. Ethics*, 1-29.
- Png, I.P., Tan, B.C. and Wee, K.L., 2001. Dimensions of national culture and corporate adoption of IT infrastructure. *IEEE Transactions on Engineering Management* 48(1), 36-45.
- Poole, W., 2004. A perspective on US international capital flows. *Federal Reserve Bank of St. Louis Review* 86, 1-8.
- Porta, R.L., Lopez-de-Silanes, F., Shleifer, A. and Vishny, R.W., 1998. Law and finance. *J. of Political Economy* 106(6), 1113-1155.
- Qu, W.G. and Yang, Z., 2015. The effect of uncertainty avoidance and social trust on supply chain collaboration. *J. of Bus. Res.* 68(5), 911-918.
- Ramirez, A. and Tadesse, S., 2009. Corporate cash holdings, uncertainty avoidance, and the multinationality of firms. *International Bus. Rev.* 18(4), p387-403.
- Rieger, M.O., Wang, M. and Hens, T., 2015. Risk preferences around the world. *Management Science* 61(3), 637-648.

- Schuler, R.S. and Rogovsky, N., 1998. Understanding compensation practice variations across firms: The impact of national culture. *J. of International Bus. Stud.* 29(1), 159-177.
- Shao, L., Kwok, C.C. and Guedhami, O., 2010. National culture and dividend policy. *J. of International Bus. Stud.* 41(8), 1391-1414.
- Soares, A., Farhangmehr, M. and Shoham, A., 2007. Hofstede's dimensions of culture in international marketing studies. *J. of Bus. Res.* 60(3), 277-284.
- Soderstrom, N. and Sun, K., 2007. IFRS adoption and accounting quality: A review. *European Account. Rev.* 16(4), 675-702.
- Stambaugh, R., Yu, J. and Yuan, Y., 2012. The short of it: Investor sentiment and anomalies. *J. of Financial Economics* 104(2), 288-302.
- Stulz, R. and Williamson, R., 2003. Culture, openness, and finance. *J. of Financial Economics* 70(3), 313-349.
- Tucker, X., and Zarowin, P., 2006. Does income smoothing improve earnings informativeness? *Account. Rev.* 81(1), 251-270.
- Van Dijk, E. and Zeelenberg, M. (2003). The discounting of ambiguous information in economic decision making, *J. of Behavioral Decis. Mak.*, 16(5), pp. 341-52.
- Warfield, T. and Wild, J., 1992. Accounting recognition and the relevance of earnings as an explanatory variable for returns. *Account. Rev.* 67(4), 821-842.
- Weetman, P., 2006. Discovering the 'international' in accounting and finance. *The Br. Account. Rev.* 38(4), 351-370.
- Williams, C. and van Triest, S., 2009. The impact of corporate and national cultures on decentralization in multinational corporations. *International Bus. Rev.* 18(2), 56-167.
- Williams, C. D. (2014). Asymmetric responses to earnings news: A case for ambiguity, *The Account. Rev.*, 90(2), pp. 785-817.
- Yip, R. W., & Young, D., 2012. Does mandatory IFRS adoption improve information comparability?. *The Account. Rev.*, 87(5), 1767-1789.
- Zeff, S.A., 2007. Some obstacles to global financial reporting comparability and convergence at a high level of quality. *The Br. Account. Rev.* 39(4), 290-302.
- Zingales, L., 2011. The "cultural revolution" in finance, *J. of Financial Economics* 117(1), 1-4.

**Table 1. Sample selection process** 

	Tuble 10 buildie beleetigii process
114,336	We focus on the countries adopting IFRS on a mandatory basis in 2005 or later and collect data from the year of IFRS adoption until 2015
26,879	Firm-year observations for which data item indicating accounting standards is missing or reporting standards are not IFRS
10,391	First-time IFRS adopters
16,326	Firms from Financial industry or unclassified
9,657	Firm-year observations that have had their financial year end changed
9,694	Firm-year observations with missing firm-specific data
11,731	Firm-year observations with missing country-specific data
1,445	Firm-year observations with missing international portfolio diversification data used to measure market openness
1,331	Outliers*
26,882	final sample [t = 2006, 2015] [4,879 firms]

<sup>\*</sup>Following Haw et al. (2012) and Lundholm and Myers (2002), we define as outliers firm-year observations when the absolute value of scaled current and lagged earnings are above one. We also define as outliers those firm-year observations when the absolute value of the sum of future earnings is above three.

Table 2. Firm-level descriptive statistics

Variable	Mean	St. Dev.	Min	Median	Max
$E_{t+1,+3}$	0.050	0.535	-2.971	0.149	1.771
$E_t$	0.011	0.169	-0.999	0.051	0.380
$E_{t-1}$	0.017	0.167	-1.000	0.047	0.584
$R_t$	0.120	0.542	-0.859	0.032	3.442
$R_{i,t+1,+3}$	0.373	1.039	-0.930	0.145	7.029
$UA * E_{t+1,+3}$	0.017	0.296	-2.655	0.028	1.678
Persistence	0.262	0.440	0.000	0.000	1.000
EarnVol	0.256	0.511	0.008	0.091	3.770
BM	0.860	0.781	-1.541	0.646	5.251
Size	12.493	2.390	5.361	12.269	20.846
Opacity	-0.036	0.088	-0.332	-0.035	0.299

See Appendix A for the definitions of the variables.

**Table 3. Country-level descriptive statistics** 

	Year of												
Country	<i>IFRS</i>	N	UA	Disclosure	Own	${\it MarketTradingVolume}$	${\it Market Volatility}$	Non Catholic	CivCom	MarketDev	InvProtection	Enforcement	Openness
	adoption												
Argentina	2012	104	86	0.949	66.784	53.066	1.584	0	1	0.010	0.342	11	6.751
Austria	2005	393	70	0.922	64.534	21.950	0.248	0	1	0.030	0.213	27	5.474
Brazil	2010	1,145	76	0.879	40.077	87.950	0.204	0	1	0.037	0.274	23	4.017
Finland	2005	885	59	0.912	32.115	83.338	0.341	1	1	0.072	0.457	32	4.965
France	2005	3,169	86	0.871	57.814	26.049	0.284	0	1	0.070	0.379	45	2.910
Germany	2005	3,471	65	0.918	53.804	0.683	0.416	0	1	0.042	0.282	44	3.135
Greece	2005	1,435	112	0.913	62.722	165.025	0.336	1	1	0.034	0.217	26	6.209
Ireland	2005	32	35	0.508	19.078	303.006	0.199	0	0	0.028	0.789	41	8.508
Israel	2008	1,379	81	0.928	66.325	248.970	2.929	1	0	0.053	0.725	48	5.361
Italy	2005	1,628	75	0.903	56.014	400.202	0.252	0	1	0.032	0.421	46	3.992
Malaysia	2012	452	36	0.910	67.001	197.274	0.290	1	0	0.110	0.950	40	4.975
Mexico	2012	94	82	0.912	61.727	134.812	0.236	0	1	0.037	0.172	25	4.758
Netherlands	2005	602	53	0.904	37.344	69.777	0.248	0	1	0.061	0.203	43	3.338
Philippines	2005	681	44	0.883	74.725	813.036	16.224	0	1	0.050	0.215	27	6.095
Portugal	2005	178	104	0.849	66.986	538.768	0.408	0	1	0.033	0.444	29	5.850
South	2005	1,185	49	0.886	41.457	79.138	0.877	1	0	0.110	0.813	29	4.009
Africa	2003	1,165	49	0.880	41.437	79.130	0.877	1	U	0.110	0.813	29	4.009
Spain	2005	685	86	0.909	54.531	104.780	0.344	0	1	0.051	0.374	42	3.630
Sweden	2005	1,952	29	0.885	29.328	101.275	0.446	1	1	0.092	0.333	34	4.097
Turkey	2005	1,859	85	0.896	64.082	1011.266	0.830	1	1	0.024	0.429	20	5.456
United	2005	5,553	35	0.898	35.830	134.066	0.812	1	0	0.112	0.950	54	2.002
Kingdom	2003	3,333	33	0.070	33.630	134.000	0.012	1	U	0.112	0.730	J <del>4</del>	2.002

See Appendix A for the definitions of the variables.

Table 4. Descriptive statistics across countries with low and high uncertainty avoidance and market openness

Panel A: Descriptive statistics across low and high UA								
	High UA	(9,268)	Low UA	(17,614)	Test o	f difference		
	Mean	Median	Mean	Median	t-test	Man-Witney test		
$E_{t+1,+3}$	0.043	0.135	0.054	0.154	-0.011*	-0.018		
$E_t$	0.013	0.050	0.010	0.051	0.003	-0.001***		
$E_{t-1}$	0.022	0.048	0.014	0.047	0.009***	0.001***		
$R_t$	0.108	0.001	0.126	0.048	-0.018***	-0.047***		
$R_{i,t+1,+3}$	0.358	0.079	0.381	0.179	-0.023**	-0.100***		
Persistence	0.265	0.000	0.261	0.000	0.004	0.000		
EarnVol	0.289	0.099	0.239	0.087	0.050***	0.012***		
BM	1.031	0.754	0.770	0.599	0.261***	0.155***		
Size	12.925	12.796	12.267	12.046	0.658***	0.750***		
Opacity	-0.029	-0.031	-0.039	-0.037	0.010***	0.006***		
Disclosure	0.898	0.905	0.898	0.907	-0.001	-0.002*		
Own	56.821	59.020	45.784	39.470	11.037***	19.550***		
MarketTradingVolume	398.746	192.223	90.859	67.736	307.887***	124.487***		
MarketVolatility	1.652	0.249	0.704	0.309	0.948***	-0.060***		
NonCatholic	0.532	1.000	0.555	1.000	-0.023***	0.000***		
CivCom	0.823	1.000	0.605	1.000	0.219***	0.000***		
MarketDev	0.048	0.035	0.077	0.070	-0.029***	-0.034***		
InvProtection	0.427	0.421	0.558	0.379	-0.131***	0.042***		
Enforcement	30.083	26.000	45.364	45.000	-15.281***	-19.000***		
Openness	4.849	4.802	3.207	3.046	1.642***	1.755***		

Panel B: Descriptive statistics across low and high market openness

	High r	narket	Low r	narket	Tost of	f difference
	openness	(18,489)	opennes	s (8,393)	Test o.	i difference
	Mean	Median	Mean	Median	t-test	Man-Witney test
$E_{t+1,+3}$	0.052	0.153	0.045	0.135	0.007	0.018
$E_t$	0.012	0.051	0.008	0.049	0.004**	0.002
$E_{t-1}$	0.018	0.048	0.014	0.045	0.004**	0.002*
$R_t$	0.107	0.035	0.147	0.027	-0.040***	0.008**
$R_{i,t+1,+3}$	0.373	0.162	0.375	0.106	-0.002	0.056***
Persistence	0.252	0.000	0.284	0.000	-0.032***	0.000***
EarnVol	0.248	0.087	0.275	0.099	-0.028***	-0.012***
BM	0.803	0.612	0.986	0.736	-0.183***	-0.123***
Size	12.523	12.286	12.428	12.234	0.095***	0.052***
Opacity	-0.038	-0.036	-0.030	-0.032	-0.008***	-0.005***
UA	57.472	65.000	75.919	81.000	-18.446***	-16.000***
Disclosure	0.894	0.897	0.906	0.913	-0.012***	-0.016***
Own	45.136	42.240	59.399	65.000	-14.263***	-22.760***
MarketTradingVolume	93.791	67.736	424.384	192.223	-330.593***	-124.487***
<i>MarketVolatility</i>	0.517	0.277	2.161	0.274	-1.644***	0.004***
NonCatholic	0.440	0.000	0.783	1.000	-0.343***	-1.000***
CivCom	0.643	1.000	0.763	1.000	-0.120***	0.000***
MarketDev	0.075	0.068	0.049	0.043	0.027***	0.025***
InvProtection	0.544	0.379	0.443	0.429	0.100***	-0.050***
Enforcement	44.226	45.000	30.996	27.000	13.231***	18.000***

Countries with high (low) uncertainty avoidance (market openness) are defined as those countries with a score above (below) the uncertainty avoidance (market openness) score of all countries in our sample. See Appendix A for the definitions of the variables. \*, \*\* and \*\*\* denote significance at the 10%, 5% and 1% levels, respectively.

Table 5. Share price anticipation of future earnings and the effect of uncertainty avoidance

(1)   (2)   (3)   (4)			avoluance	Stock returns at year t	
Constant         Base line         Uncertainty avoidance value         Firm level controls         Firm and country controls           Constant $0.416^{****}$ $0.412^{****}$ $0.475^{****}$ $0.351^{****}$ $E_{r+1,+3}$ $0.169^{****}$ $0.203^{****}$ $0.260^{****}$ $0.301^{****}$ $E_t$ $0.598^{*****}$ $0.518^{*****}$ $0.505^{****}$ $E_t$ $0.417^{*****}$ $0.420^{*****}$ $0.518^{****}$ $0.505^{*****}$ $E_t$ $0.417^{******}$ $0.420^{**********}$ $0.518^{******}$ $0.505^{*******}$ $E_t$ $0.417^{************************************$		(1)			(4)
Constant $0.416^{****}$ $0.412^{****}$ $0.475^{****}$ $0.351^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{****}$ $0.415^{*****}$ $0.301^{****}$ $0.169^{****}$ $0.203^{****}$ $0.203^{****}$ $0.260^{****}$ $0.301^{****}$ $0.508^{*****}$ $0.508^{*****}$ $0.508^{*****}$ $0.508^{*****}$ $0.508^{*****}$ $0.508^{****}$ $0.508^{****}$ $0.508^{****}$ $0.518^{****}$ $0.508^{****}$ $0.508^{****}$ $0.538^{****}$ $0.538^{****}$ $0.538^{****}$ $0.538^{****}$ $0.538^{****}$ $0.338^{****}$ $0.338^{****}$ $0.338^{****}$ $0.338^{****}$ $0.338^{****}$ $0.338^{****}$ $0.033^{****}$ $0.033^{****}$ $0.033^{****}$ $0.003^{****}$ $0.003^{****}$ $0.003^{****}$ $0.003^{****}$ $0.0019^{*****}$ $0.0019^{*****}$ $0.0019^{*******}$ $0.0019^{**********}$ $0.0019^{************************************$			Uncertainty		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Constant				
$E_{I} = (19.73)  (14.62)  (8.02)  (5.75) \\ E_{I} = (0.598*** 0.597*** 0.518*** 0.505*** \\ (24.36)  (24.33)  (16.69)  (16.66) \\ E_{IJ} = (-0.417*** - 0.420*** - 0.395*** - 0.335*** \\ (-15.93)  (-16.04)  (-15.32)  (-14.15) \\ R_{I+I,+3} = (-0.033*** - 0.033*** - 0.029*** - 0.032*** \\ (-8.94)  (-8.97)  (-7.77)  (-8.87) \\ UA*E_{I+I,+3} = (-0.049*** - 0.081*** - 0.013*** \\ (-3.46)  (-3.98)  (-2.89) \\ UA = (-10.13)  (0.19* - 0.173*** - 0.103*** \\ (-10.74)  0.013  0.019*  -0.173*** \\ (-1.158)  (-10.74) \\ 0.002ity*E_{I+I,+3} = (-1.62)  0.069*** - 0.069*** \\ 0.006*** - 0.069*** - 0.069*** - 0.069*** \\ 0.006*** - 0.069*** - 0.069*** - 0.069*** \\ 0.006*** - 0.059*** - 0.069*** - 0.069*** \\ 0.006*** - 0.059*** - 0.091*** \\ 0.062*** - 0.069*** - 0.091*** \\ 0.062*** - 0.069*** - 0.091*** \\ 0.062*** - 0.069*** - 0.091*** \\ 0.062*** - 0.069*** - 0.091*** \\ 0.068*** - 0.072*** - 0.091*** \\ 0.068*** - 0.072*** - 0.091*** \\ 0.068*** - 0.072*** - 0.002 - 0.009 \\ 0.006  (-0.35)  0.091*** \\ 0.011*** - 0.002 - 0.009 \\ 0.006  (-0.35)  0.004** - 0.002 \\ 0.009  0.009  0.009 \\ 0.001*** - 0.001** - 0.003 \\ 0.010*** - 0.001** - 0.003 \\ 0.010*** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0.003 \\ 0.021** - 0$	_		` /		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$E_{t+1,+3}$				
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	_	, ,			
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$E_t$				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	_	, ,	, ,	, ,	` /
$R_{it+1,+3} = -0.033^{***} - 0.033^{***} - 0.033^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.069^{***} - 0.081^{***} - 0.013^{***} - 0.013^{***} - 0.069^{***} - 0.081^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.013^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.032^{***} - 0.060^{***} -$	$E_{t-1}$				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		'	,	,	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$R_{i,t+1,+3}$				
$ \begin{array}{c} (-3.46) & (-3.98) & (-2.89) \\ 0.013 & 0.019 & -0.173*** \\ (1.21) & (1.80) & (-10.74) \\ 0.032 & -0.032 \\ (-1.58) & (-1.62) \\ 0.060*** & -0.060*** \\ (-5.79) & (-5.99) \\ 0.060*** & 0.075**** \\ (-5.79) & (-5.99) \\ 0.088*** & 0.075**** \\ (-5.99) & (-0.06) \\ 0.062*** & 0.062*** & 0.059*** \\ (-0.02) & (-0.02) \\ 0.022 & (-0.02) \\ 0.022 & (-0.02) \\ 0.022 & (-0.02) \\ 0.032 & (-0.03) \\ 0.062*** & 0.059*** \\ 0.062*** & 0.059*** \\ 0.062*** & 0.059*** \\ 0.062*** & 0.059*** \\ 0.060*** & -0.091*** \\ 0.061** & -0.091*** \\ 0.061** & -0.091*** \\ 0.061** & -0.091*** \\ 0.068*** & -0.072*** \\ 0.068*** & -0.072*** \\ 0.066** & (-6.43) & (-6.89) \\ 0.002 & -0.009 \\ 0.003 & (-0.06) & (-0.35) \\ 0.110*** & 0.105*** \\ 0.955) & (-9.40) \\ 0.110*** & 0.105*** \\ 0.955) & (-9.40) \\ 0.166*** & -0.041* & -0.066*** \\ 0.127*** & -0.027*** \\ 0.027*** & -0.227*** \\ 0.027*** & -0.227*** \\ 0.031 & (-0.13) \\ 0.001 & 0.010*** \\ 0.011*** \\ 0.013 & (-0.13) \\ 0.004 & (-0.13) \\ 0.076*** & (5.97) \\ 0.0044** & 0.076*** \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040*** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.021) \\ 0.0040** & 0.021 \\ 0.0$		(-8.94)			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$UA*E_{t+1,+3}$				
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	UA				
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$			(1.21)		, ,
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$Opacity*E_{t+1,+3}$				-0.032
$\begin{array}{cccccccccccccccccccccccccccccccccccc$				` ,	
$\begin{array}{c} Size_{t} * E_{t+1,+3} & 0.088*** & 0.075*** \\ (3.96) & (3.40) \\ Size_{t} & 0.062*** & 0.059*** \\ (6.02) & (6.02) \\ (6.02) & (6.02) \\ (6.02) & (6.02) \\ \end{array}$ $\begin{array}{c} Persistence * E_{t+1,+3} & -0.089*** & -0.091*** \\ (-6.14) & (-6.40) \\ -0.068*** & -0.072*** \\ (-6.43) & (-6.89) \\ \end{array}$ $\begin{array}{c} EarnVol^* E_{t+1,+3} & -0.002 & -0.009 \\ (-0.06) & (-0.35) \\ \end{array}$ $\begin{array}{c} EarnVol & 0.110*** & 0.105*** \\ (9.55) & (9.40) \\ \end{array}$ $\begin{array}{c} BM_{t} * E_{t+1,+3} & -0.041* & -0.066*** \\ (-1.92) & (-3.16) \\ \end{array}$ $\begin{array}{c} BM_{t} & -0.227*** & -0.227*** \\ (-20.88) & (-21.60) \\ \end{array}$ $\begin{array}{c} Disclosure * E_{t+1,+3} & -0.003 \\ (-0.12) \\ \end{array}$ $\begin{array}{c} C.003 \\ (-0.13) \\ C.011*** \\ \end{array}$ $\begin{array}{c} C.0040*** \\ (-0.13) \\ C.011*** \\ \end{array}$ $\begin{array}{c} C.0040*** \\ (-0.13) \\ C.011*** \\ \end{array}$ $\begin{array}{c} C.0.031 \\ C.1.17 \\ \end{array}$ $\begin{array}{c} MarketTradingVolume * \\ E_{t+1,+3} & 0.004 \\ (-0.13) \\ C.0.076*** \\ \end{array}$ $\begin{array}{c} C.9.7 \\ C.9.2 \\ C.9.7 \\ C.9.7 \\ C.9.7 \\ C.9.7 \\ C.9.3 \\ C.9.7 \\ $	Opacity			-0.060***	-0.060***
Sizet $(3.96)$ $(3.40)$ Sizet $0.062^{***}$ $0.059^{***}$ $(6.02)$ $(6.02)$ $(6.02)$ Persistence * $E_{t+1,+3}$ $-0.089^{***}$ $-0.091^{****}$ $(-6.14)$ $(-6.40)$ $(-6.40)$ Persistence $-0.068^{***}$ $-0.072^{****}$ $(-6.43)$ $(-6.89)$ $(-6.89)$ EarnVol** $E_{t+1,+3}$ $-0.002$ $-0.009$ $(-0.6)$ $(-0.35)$ $(-0.06)$ $EarnVol$ $0.110^{***}$ $0.105^{****}$ $EarnVol$ $0.110^{****}$ $0.105^{****}$ $EarnVol$ $0.110^{****}$ $0.105^{****}$ $EarnVol$ $0.105^{****}$ $0.066^{****}$ $(-0.55)$ $(9.40)$ $0.105^{****}$ $EarnVol$ $0.101^{****}$ $0.066^{****}$ $(-9.55)$ $(9.40)$ $0.066^{****}$ $(-1.92)$ $(-3.16)$ $0.003$ $(-2.27^{*****}$ $0.0227^{*******************         (-2.088) (-2.1.60) 0.004 (-3.64) 0.004 0.004 (-3.64) 0.004$				` ,	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$Size_t*E_{t+1,+3}$			0.088***	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$				(3.96)	(3.40)
$\begin{array}{llllllllllllllllllllllllllllllllllll$	$Size_t$			0.062***	0.059***
$\begin{array}{ccccccccccccccc} & (-6.14) & (-6.40) \\ Persistence & -0.068*** & -0.072*** \\ & (-6.43) & (-6.89) \\ EarnVol* E_{t+l,+3} & -0.002 & -0.009 \\ & (-0.06) & (-0.35) \\ EarnVol & 0.110*** & 0.105*** \\ & (9.55) & (9.40) \\ BM_i* E_{t+l,+3} & -0.041* & -0.066*** \\ & (-1.92) & (-3.16) \\ & (-2.27*** & -0.227*** \\ & (-20.88) & (-21.60) \\ Disclosure* E_{t+l,+3} & & -0.003 \\ & (-0.12) \\ Disclosure & & -0.040*** \\ & (-3.64) \\ Own* E_{t+l,+3} & & & (-0.10) \\ Own & & & & (-1.30) \\ Own & & & & (-1.7) \\ MarketTradingVolume* & & & & (-1.17) \\ MarketTradingVolume & & & & (5.97) \\ MarketVolatility* E_{t+l,+3} & & & 0.004 \\ & & & & & (0.21) \\ MarketVolatility & (0.21) \\ Ma$				(6.02)	(6.02)
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Persistence * $E_{t+1,+3}$			-0.089***	-0.091***
$EarnVol* E_{t+l,+3}$ (-6.43)       (-6.89) $EarnVol$ $-0.002$ $-0.009$ $EarnVol$ $0.110***$ $0.105***$ $EarnVol$ $0.110***$ $0.105***$ $EarnVol$ $0.110***$ $0.105***$ $EarnVol$ $0.110***$ $0.066***$ $EarnVol$ $0.041*$ $0.066***$ $EarnVol$ $0.041*$ $0.066***$ $EarnVol$ $0.041*$ $0.066***$ $EarnVol$ $0.041*$ $0.066***$ $EarnVol$ $0.027****$ $0.003$ $EarnVol$ $0.041*$ $0.041*$ $EarnVol$ $0.041*$ $0.004*$ $EarnVol$ $0.041*$ $0.004*$ $EarnVol$ $0.004*$ $0.004*$ $EarnVol$ $0.004*$ $0.004*$ $EarnVol$ $0.004*$ $0.004*$ $EarnVol$ $0.004*$ <				(-6.14)	(-6.40)
$EarnVol* E_{t+1,+3}$ $-0.002$ $-0.009$ $EarnVol$ $(-0.06)$ $(-0.35)$ $BM_t* E_{t+1,+3}$ $0.10****$ $0.105****$ $BM_t* E_{t+1,+3}$ $-0.041*$ $-0.066***$ $BM_t$ $-0.227***$ $-0.227***$ $(-2.0.88)$ $(-21.60)$ $Disclosure* E_{t+1,+3}$ $-0.003$ $(-0.12)$ $Disclosure$ $-0.040****$ $(-0.13)$ $Own* E_{t+1,+3}$ $-0.004$ $(-0.13)$ $Own$ $0.101****$ $(-0.13)$ $Own$ $0.101***$ $(-0.03)$ $E_{t+1,+3}$ $-0.031$ $(-1.17)$ $MarketTradingVolume*$ $0.076***$ $(5.97)$ $MarketVolatility* E_{t+1,+3}$ $0.004$ $(0.21)$ $MarketVolatility$ $0.344***$	Persistence			-0.068***	-0.072***
$EarnVol$ $(-0.06)$ $(-0.35)$ $BM_l^*E_{l+l,+3}$ $(-9.55)$ $(9.40)$ $BM_l^*E_{l+l,+3}$ $-0.041^*$ $-0.066^{***}$ $BM_l$ $-0.227^{****}$ $-0.227^{****}$ $BM_l$ $-0.227^{****}$ $-0.227^{****}$ $C_l^*B_l^*$ $(-20.88)$ $(-21.60)$ $Disclosure^*E_{l+l,+3}$ $-0.003$ $(-0.12)$ $Disclosure$ $(-0.12)$ $(-0.12)$ $Disclosure$ $(-0.04)$ $(-0.13)$ $Own^*E_{l+l,+3}$ $-0.004$ $(-0.13)$ $Own$ $0.101^{***}$ $(-0.13)$ $Own$ $0.101^{***}$ $(-0.03)$ $E_{l+l,+3}$ $(-0.03)$ $(-1.17)$ $MarketTradingVolume$ $0.076^{***}$ $(-0.076^{***})$ $MarketVolatility^*E_{l+l,+3}$ $0.004$ $(0.21)$ $MarketVolatility$ $0.344^{***}$				(-6.43)	(-6.89)
$EarnVol$ 0.110***       0.105*** $(9.55)$ $(9.40)$ $BM_i * E_{t+1,+3}$ $-0.041*$ $-0.066***$ $(-1.92)$ $(-3.16)$ $BM_t$ $-0.227***$ $-0.227***$ $(-20.88)$ $(-21.60)$ $Disclosure * E_{t+1,+3}$ $-0.003$ $(-0.12)$ $-0.040***$ $(-3.64)$ $-0.004$ $(-0.13)$ $-0.004$ $(-0.13)$ $-0.001$ $Column * E_{t+1,+3}$ $-0.031$ $Column * E_{t+1,+3}$ $-0.031$ $Column * E_{t+1,+3}$ $-0.031$ $Column * E_{t+1,+3}$ $-0.003$ $Column * E_{t+1,+3}$ <t< td=""><td><math>EarnVol*E_{t+1,+3}</math></td><td></td><td></td><td>-0.002</td><td>-0.009</td></t<>	$EarnVol*E_{t+1,+3}$			-0.002	-0.009
$EarnVol$ 0.110***       0.105*** $(9.55)$ $(9.40)$ $BM_i * E_{t+1,+3}$ $-0.041*$ $-0.066***$ $(-1.92)$ $(-3.16)$ $BM_t$ $-0.227***$ $-0.227***$ $(-20.88)$ $(-21.60)$ $Disclosure * E_{t+1,+3}$ $-0.003$ $(-0.12)$ $-0.040***$ $(-3.64)$ $-0.004$ $(-0.13)$ $-0.004$ $(-0.13)$ $-0.001$ $Column * E_{t+1,+3}$ $-0.031$ $Column * E_{t+1,+3}$ $-0.031$ $Column * E_{t+1,+3}$ $-0.031$ $Column * E_{t+1,+3}$ $-0.003$ $Column * E_{t+1,+3}$ <t< td=""><td></td><td></td><td></td><td>(-0.06)</td><td>(-0.35)</td></t<>				(-0.06)	(-0.35)
$BM_t * E_{t+1,+3}$ $-0.041 *$ $-0.066 ***$ $BM_t$ $-0.227 ***$ $-0.227 ***$ $C = 0.88$ $C = 0.003$ $C = 0.003$ $C = 0.12$ $C = 0.003$ $C = 0.003$ $C = 0.003$ $C = 0.004$	EarnVol			0.110***	
$BM_{l} \qquad (-1.92) \qquad (-3.16) \\ -0.227*** \qquad -0.227*** \\ (-20.88) \qquad (-21.60) \\ Disclosure* E_{l+l,+3} \qquad \qquad -0.003 \\ (-0.12) \\ Disclosure \qquad \qquad -0.040*** \\ (-3.64) \\ Own* E_{l+l,+3} \qquad \qquad -0.004 \\ (-0.13) \\ Own \qquad \qquad 0.101*** \\ (-1.17) \\ Market Trading Volume* \\ E_{l+l,+3} \qquad \qquad -0.031 \\ E_{l+l,+3} \qquad \qquad (-1.17) \\ Market Trading Volume \qquad \qquad 0.076*** \\ (5.97) \\ Market Volatility* E_{l+l,+3} \qquad \qquad 0.004 \\ (0.21) \\ Market Volatility \qquad \qquad 0.344*** \\ \\ Market Volatility \qquad \qquad 0.344*** \\ \\ \\ Darket Volatility \qquad \qquad 0.344*** \\ \\ Darket Volatility \qquad \qquad 0.344** \\ \\ Darket Volatility \qquad \qquad 0.344*** \\ \\ Darket Volatility \qquad \qquad 0.344** \\ $				(9.55)	(9.40)
$BM_{l} \qquad (-1.92) \qquad (-3.16) \\ -0.227*** \qquad -0.227*** \\ (-20.88) \qquad (-21.60) \\ Disclosure* E_{l+l,+3} \qquad \qquad -0.003 \\ (-0.12) \\ Disclosure \qquad \qquad -0.040*** \\ (-3.64) \\ Own* E_{l+l,+3} \qquad \qquad -0.004 \\ (-0.13) \\ Own \qquad \qquad 0.101*** \\ (-1.17) \\ Market Trading Volume* \\ E_{l+l,+3} \qquad \qquad -0.031 \\ E_{l+l,+3} \qquad \qquad (-1.17) \\ Market Trading Volume \qquad \qquad 0.076*** \\ (5.97) \\ Market Volatility* E_{l+l,+3} \qquad \qquad 0.004 \\ (0.21) \\ Market Volatility \qquad \qquad 0.344*** \\ \\ Market Volatility \qquad \qquad 0.344*** \\ \\ \\ Darket Volatility \qquad \qquad 0.344*** \\ \\ Darket Volatility \qquad \qquad 0.344** \\ \\ Darket Volatility \qquad \qquad 0.344*** \\ \\ Darket Volatility \qquad \qquad 0.344** \\ $	$BM_t*E_{t+1.+3}$			-0.041*	-0.066***
$BM_{t}$ $-0.227***$ $-0.227***$ $(-20.88)$ $(-21.60)$ $Disclosure* E_{t+1,+3}$ $-0.003$ $(-0.12)$ $Disclosure$ $-0.040***$ $(-3.64)$ $Own* E_{t+1,+3}$ $-0.004$ $(-0.13)$ $Own$ $0.101***$ $(-1.17)$ $MarketTradingVolume*$ $-0.031$ $(-1.17)$ $MarketTradingVolume$ $0.076***$ $(5.97)$ $MarketVolatility* E_{t+1,+3}$ $0.004$ $(0.21)$ $MarketVolatility$ $0.344***$					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$BM_t$				
Disclosure * $E_{t+1,+3}$ -0.003         Disclosure       -0.040***         -0.004       (-3.64)         Own * $E_{t+1,+3}$ -0.004         Own       0.101***         MarketTradingVolume*       -0.031 $E_{t+1,+3}$ (-1.17)         MarketTradingVolume       0.076***         MarketVolatility* $E_{t+1,+3}$ 0.004         MarketVolatility       0.344***	•			(-20.88)	(-21.60)
Disclosure $(-0.12)$ $-0.040***$ $(-3.64)$ $Own* E_{t+I,+3}$ $(-0.004)$ $Own$ $(-0.13)$ $Own$ $(-0.13)$ $MarketTradingVolume*$ $(-0.031)$ $E_{t+I,+3}$ $(-1.17)$ $MarketTradingVolume$ $(-0.076***$ $(5.97)$ $(0.004)$ $MarketVolatility* E_{t+I,+3} (0.21) MarketVolatility (0.344****) $	$Disclosure * E_{t+1+3}$			,	
Disclosure $-0.040^{***}$ $Own^* E_{t+I,+3}$ $-0.004$ $Own$ $0.101^{***}$ $Own$ $0.101^{***}$ $MarketTradingVolume^*$ $-0.031$ $E_{t+I,+3}$ $(-1.17)$ $MarketTradingVolume$ $0.076^{***}$ $(5.97)$ $0.004$ $MarketVolatility^* E_{t+I,+3}$ $0.004$ $MarketVolatility$ $0.344^{***}$	, , , , , , , , , , , , , , , , , , ,				
$Own*E_{t+I,+3}$ $(-3.64)$ $Own$ $(-0.13)$ $Own$ $0.101***$ $MarketTradingVolume*$ $(-1.17)$ $E_{t+I,+3}$ $(-1.17)$ $MarketTradingVolume$ $(5.97)$ $MarketVolatility*E_{t+I,+3}$ $0.004$ $MarketVolatility$ $(0.21)$ $MarketVolatility$ $0.344****$	Disclosure				
$Own*E_{t+1,+3}$ $-0.004$ $Own$ $0.101***$ $MarketTradingVolume*$ $-0.031$ $E_{t+1,+3}$ $(-1.17)$ $MarketTradingVolume$ $0.076***$ $MarketVolatility*E_{t+1,+3}$ $0.004$ $MarketVolatility$ $0.344***$					
Own $(-0.13)$ $0.101***$ $(7.19)$ MarketTradingVolume* $-0.031$ MarketTradingVolume $(-1.17)$ MarketVolatility* $E_{t+1,+3}$ $(5.97)$ MarketVolatility $(0.21)$ MarketVolatility $(0.344***$	$Own^*E_{t+1+3}$				
$Own$ $0.101***$ $MarketTradingVolume*$ $-0.031$ $E_{t+1,+3}$ $(-1.17)$ $MarketTradingVolume$ $0.076***$ $MarketVolatility* E_{t+1,+3}$ $0.004$ $MarketVolatility$ $0.344***$	=111,13				
$MarketTradingVolume*$ (7.19) $E_{t+1,+3}$ -0.031 $MarketTradingVolume$ (-1.17) $MarketVolatility*E_{t+1,+3}$ (5.97) $MarketVolatility*E_{t+1,+3}$ 0.004 $MarketVolatility$ 0.344***	Own				
$MarketTradingVolume*$ -0.031 $E_{t+1,+3}$ (-1.17) $MarketTradingVolume$ 0.076*** $(5.97)$ (5.97) $MarketVolatility* E_{t+1,+3}$ 0.004 $MarketVolatility$ 0.344***	· · · · ·				
$E_{t+1,+3}$ -0.031         MarketTradingVolume $(-1.17)$ MarketVolatility* $E_{t+1,+3}$ $(5.97)$ MarketVolatility $(0.21)$ MarketVolatility $(0.344****)$	MarketTradinoVolume*				
MarketTradingVolume $(-1.17)$ MarketVolatility* $E_{t+1,+3}$ $(5.97)$ MarketVolatility $(0.21)$ MarketVolatility $(0.344***)$	_				-0.031
MarketTradingVolume $0.076***$ $(5.97)$ $(5.97)$ MarketVolatility* $E_{t+1,+3}$ $0.004$ $(0.21)$ $(0.344***)$	— i + 1, ₹J				(-1.17)
$ \begin{array}{c} (5.97) \\ \textit{MarketVolatility*} E_{t+1,+3} & 0.004 \\ (0.21) \\ \textit{MarketVolatility} & 0.344*** \end{array} $	MarketTradingVolume				
MarketVolatility* $E_{t+1,+3}$ 0.004MarketVolatility(0.21)MarketVolatility0.344***					
(0.21) MarketVolatility 0.344***	MarketVolatility* F				
MarketVolatility 0.344***	11101 100 1 0 100 101 1 y L[+1,+3				
•	MarketVolatility				
(3/10)	man nei rounniny				(32.06)

$NonCatholic*E_{t+1,+3}$				-0.001
				(-0.07)
NonCatholic				-0.087***
				(-8.64)
$CivCom^*E_{t+1,+3}$				0.007
				(0.21)
CivCom				0.002
				(0.13)
$MarketDev*E_{t+1,+3}$				0.062*
				(1.70)
MarketDev				0.047***
				(2.58)
$InvProtection*E_{t+1,+3}$				-0.039
				(-0.90)
InvProtection				0.072***
				(3.70)
Enforcement* $E_{t+1,+3}$				-0.021
				(-0.58)
Enforcement				-0.149***
				(-8.59)
Year FEs	Yes	Yes	Yes	Yes
Industry FEs	Yes	Yes	Yes	Yes
N	26,882	26,882	26,882	26,882
R2-Adj	0.261	0.262	0.284	0.316
F	412.6	378.3	316.8	218.6

This table shows the significance of uncertainty avoidance regarding the relation between current stock returns and firm's future earnings. Our dependent variable is stock returns at year t and the main independent variable is the interaction between uncertainty avoidance and the sum of future earnings of firm i for the 3 years following the current year ( $UA*E_{t+1,+3}$ ). Appendix A offers detailed definitions of the variables used. Standard errors are clustered at the firm level. T-statistics are shown in parentheses. \*\*\*, \*\* and \* indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Table 6. Share price anticipation of future earnings, uncertainty avoidance and market openness

	(1)	eturns at year t (2)
	Firm, country and market openness controls	Firm, country controls and the moderating effect of market openness
Constant	0.353***	0.312***
$E_{t+I,+3}$	(9.77) 0.335***	(8.53) 0.379***
$E_t$	(5.59) 0.504***	(6.13) 0.504***
$E_{t-1}$	(16.66) -0.358*** (-14.15)	(16.65) -0.358*** (-14.20)
$R_{i,t+I,+3}$	-0.032*** (-8.84)	-0.033*** (-9.17)
$UA*E_{t+1,+3}$	-0.103*** (-2.86)	-0.206*** (-3.64)
UA	-0.173*** (-10.70)	-0.063** (-2.14)
$Openness * E_{t+1,+3}*UA$		0.255** (2.45)
$Openness * E_{t+1,+3}$	-0.058 (-1.18)	-0.130** (-2.30)
Openness *UA		-0.241*** (-4.56)
Openness	-0.003 (-0.15)	0.067*** (2.64)
Opacity* $E_{t+1,+3}$	-0.031 (-1.55)	-0.031 (-1.58)
Opacity	-0.060*** (-6.02)	-0.060*** (-6.05)
$Size_t * E_{t+1,+3}$	0.076*** (3.43)	0.077*** (3.46)
$Size_t$	0.059*** (6.03)	0.059*** (6.00)
Persistence * $E_{t+1,+3}$	-0.091*** (-6.39)	-0.092*** (-6.44)
Persistence	-0.072*** (-6.88)	-0.073*** (-6.98)
$EarnVol*E_{t+1,+3}$	-0.011 (-0.44)	-0.018 (-0.69)
EarnVol	0.105*** (9.43)	0.107*** (9.64)
$BM_t * E_{t+I,+3}$ $BM_t$	-0.066*** (-3.12) -0.227***	-0.067*** (-3.20) -0.226***
$Disclosure * E_{t+1,+3}$	-0.227 -0.008 (-0.38)	-0.226*** -0.008 (-0.34)
Disclosure E <sub>t+1,+3</sub>	-0.38) -0.040*** (-3.76)	-0.039*** (-3.64)
$Own^*E_{t+L+3}$	-0.030 (-0.86)	-0.007 (-0.20)
$Own$ $D_{i+1,+3}$	0.100*** (6.44)	0.20) 0.078*** (4.68)
MarketTradingVolume* $E_{t+1,+3}$	-0.045 (-1.50)	-0.059** (-1.98)

MarketTradingVolume $(5.14)$ $0.010$ $(5.54)$ $0.014$ MarketVolatility* $E_{t+1,+3}$ $(0.43)$ $0.344***$ $(0.62)$ $0.336***$ MarketVolatility $(31.22)$ $-0.016$ $(30.70)$ $0.029$ NonCatholic* $E_{t+1,+3}$ $(-0.67)$ $-0.087***$ $(0.98)$ $-0.120***$ NonCatholic $(-6.99)$ $0.005$ $(-7.67)$ $0.002$ CivCom* $E_{t+1,+3}$ $(0.14)$ $0.001$ $(0.06)$ $-0.004$ CivCom $(0.06)$ $0.067*$ $(-0.32)$ $0.067*$ MarketDev* $E_{t+1,+3}$ $(1.82)$ $(0.87)$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
MarketVolatility $(31.22)$ $-0.016$ $(30.70)$ $0.029$ NonCatholic* $E_{t+I,+3}$ $(-0.67)$ $-0.087***$ $(0.98)$ $-0.120***$ NonCatholic $(-6.99)$ $0.005$ $(-7.67)$ $0.002$ CivCom* $E_{t+I,+3}$ $(0.14)$ $0.001$ $(0.06)$ $-0.004$ CivCom $(0.06)$ $0.067*$ $(-0.32)$ $0.033$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
NonCatholic* $E_{t+1,+3}$ (-0.67) -0.087***(0.98) -0.120***NonCatholic(-6.99) 0.005(-7.67) $0.005$ 0.002 $CivCom^* E_{t+1,+3}$ (0.14) 0.001(0.06) -0.004 $CivCom$ (0.06) (-0.32) 0.067*(-0.32) 0.033
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
NonCatholic $(-6.99)$ $(-7.67)$ $0.005$ $0.002$ $CivCom^* E_{t+I,+3}$ $(0.14)$ $(0.06)$ $0.001$ $-0.004$ $CivCom$ $(0.06)$ $(-0.32)$ $0.067^*$ $0.033$
$\begin{array}{cccc} & 0.005 & 0.002 \\ CivCom^* E_{t+I,+3} & (0.14) & (0.06) \\ & 0.001 & -0.004 \\ CivCom & (0.06) & (-0.32) \\ & 0.067^* & 0.033 \end{array}$
$CivCom^* E_{t+I,+3}$ $(0.14)$ $(0.06)$ $0.001$ $-0.004$ $CivCom$ $(0.06)$ $(-0.32)$ $0.067^*$ $0.033$
0.001       -0.004         CivCom       (0.06)       (-0.32)         0.067*       0.033
CivCom (0.06) (-0.32) 0.067* 0.033
0.067* 0.033
$Market Dev * F \qquad (1.82)$
(0.07)
0.046** 0.072***
MarketDev (2.49) (3.57)
-0.013 -0.077
$InvProtection* E_{t+1,+3} \qquad (-0.26)$
0.074*** 0.138***
InvProtection  (3.44)
-0.003 0.032
Enforcement* $E_{t+1,+3}$ (-0.09)
-0.147*** -0.173***
Enforcement $(-8.09)$ $(-9.00)$
(-21.57) (-21.54)
Year FEs Yes Yes
Industry FEs Yes Yes
N 26,882 26,882
R2-Adj 0.32 0.32
F 211.4 205.9

This table shows the significance of uncertainty avoidance and market openness regarding the relation between current stock returns and firm's future earnings. Our dependent variable is stock returns at year t and the main independent variable is the interaction between uncertainty avoidance and future earnings ( $UA*E_{t+I,+3}$ ) and the interaction between uncertainty avoidance, future earnings and market openness ( $Openness*E_{t+I,+3}*UA$ ). Appendix A offers detailed definitions of the variables used. Standard errors are clustered at the firm level. T-statistics are shown in parentheses. \*\*\*, \*\* and \* indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Table 7. Share price anticipation of future earnings, uncertainty avoidance and market openness: Additional cultural factors

	Stoc	k returns at year t
	(1)	(2)
	Firm and country controls	Firm and country controls and the
		moderating effect of market openness
Constant	0.361***	0.311***
	(10.51)	(8.45)
$E_{t+1,+3}$	0.264***	0.375***
	(4.96)	(6.11)
$E_t$	0.507***	0.507***
	(16.72)	(16.76)
$E_{t-1}$	-0.357***	-0.357***
	(-14.12)	(-14.14)
$R_{i,t+1,+3}$	-0.032***	-0.033***
,,,,,,,	(-8.91)	(-9.24)
$UA*E_{t+1,+3}$	-0.133***	-0.294***
	(-3.04)	(-4.87)
UA	-0.148***	-0.034
	(-7.52)	(-1.12)
Openness *	(7.62)	0.377***
$E_{t+1,+3}$ *UA		0.377
		(3.45)
Openness *UA		-0.251***
		(-4.74)
Openness * $E_{t+1,+3}$		-0.228***
		(-3.70)
Openness		0.098***
o p emicos		(3.57)
$PD*E_{t+1,+3}$	0.094**	0.163***
$\mathcal{L}_{l+1,+3}$	(2.54)	
PD	-0.012	(4.01) -0.035**
FD		
	(-0.71)	(-2.13)
$IDV*E_{t+1,+3}$	0.104*	0.146**
	(1.92)	(2.57)
IDV	-0.075***	-0.083***
	(-2.86)	(-2.76)
Firm controls	Yes	Yes
Country controls	Yes	Yes
Year FEs	Yes	Yes
Industry FEs	Yes	Yes
N	26,882	26,882
R2-Adj	0.316	0.317
F	202.7	191.8

This table shows the significance of uncertainty avoidance and market openness regarding the relation between current stock returns and firm's future earnings when controlling for additional cultural factors. Our dependent variable is stock returns at year t and the main independent variable is the interaction between uncertainty avoidance and future earnings ( $UA*E_{t+1,+3}$ ) and the interaction between uncertainty avoidance, future earnings and market openness ( $Openness*E_{t+1,+3}*UA$ ). Appendix A offers detailed definitions of the variables used. Standard errors are clustered at the firm level. T-statistics are shown in parentheses. \*\*\*, \*\* and \* indicate statistical significance at the 1%, 5% and 10% levels, respectively.

**Table 8. Robustness tests** 

		ries with fewer		ede to measure	Exports to
		bservations		y avoidance	GDP
	(1)	(2)	(3)	y avoidance (4)	(5)
Constant	0.345***	0.298***	0.314***	0.354***	0.356***
Constant	(10.27)	(8.19)	(9.74)	(10.19)	(7.73)
$UA * E_{t+1,+3}$	-0.098***	-0.198***	-0.104***	-0.202***	-0.155**
$D_{l+1,+3}$	(-2.82)	(-3.64)	(-3.34)	(-4.72)	(-2.04)
$E_{t+1}$	0.297***	0.377***	0.320***	0.412***	0.400***
21+1	(5.98)	(6.36)	(6.49)	(6.91)	(4.88)
$E_t$	0.492***	0.492***	0.499***	0.499***	0.490***
-1	(16.32)	(16.33)	(17.35)	(17.36)	(16.62)
$E_{t-1}$	-0.356***	-0.355***	-0.372***	-0.373***	-0.369***
÷ ÷	(-14.11)	(-14.12)	(-15.38)	(-15.43)	(-14.94)
$R_{i,t+1,+3}$	-0.037***	-0.038***	-0.031***	-0.031***	-0.031***
eye : 29 1 50	(-10.52)	(-10.76)	(-8.80)	(-8.79)	(-8.93)
UA	-0.150***	-0.070**	-0.095***	-0.136***	-0.181***
	(-9.69)	(-2.57)	(-6.75)	(-6.63)	(-5.17)
Openness *	` /	` /	` /	` /	` /
$E_{t+1,+3}$ *UA		0.246**		0.235***	
, ,		(2.50)		(2.92)	
Openness * $E_{t+1,+3}$		-0.130**		-0.138***	
•		(-2.34)		(-2.80)	
Openness *UA		-0.180***		0.074**	
1		(-3.59)		(2.11)	
Openness		0.077***		-0.060***	
· 1		(3.06)		(-2.75)	
$Trade * E_{t+1,+3}*UA$		, ,		` /	0.240*
7.1,.0					(1.71)
<i>Trade</i> * $E_{t+1,+3}$					-0.075
,					(-1.03)
Trade *UA					0.154**
					(2.28)
Trade					-0.056
					(-1.63)
Firm controls	Yes	Yes	Yes	Yes	Yes
Country controls	Yes	Yes	Yes	Yes	Yes
Year FEs	Yes	Yes	Yes	Yes	Yes
Industry FEs	Yes	Yes	Yes	Yes	Yes
maasay 1 Ls	103	103	103	103	103
N	26,652	26,652	29,144	29,144	28,242
R2-Adj	0.319	0.319	0.310	0.310	0.317
F	218.7	205.8	226.4	211.7	216.6

This table shows the robustness of results reported in Table 5 and Table 6. Once again, our dependent variable is stock returns at year t and the main independent variable is the interaction between secrecy and earnings. All controls used earlier are added at all estimations, though due to space considerations, we only present the most important parameter coefficients. Appendix A offers detailed definitions of the variables used. Standard errors are clustered at the firm level. T-statistics are shown in parentheses. \*\*\*, \*\* and \* indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Table 9. Share price anticipation of future earnings, uncertainty avoidance and market openness: Aggregate results

	Future earnings response coefficient						
		Full sample	•	Excluding UK			
	(1)	(2)	(3)	(4)			
	Country- level controls	Country-level controls and moderating effect of market openness	Country- level controls	Country-level controls and moderating effect of market openness			
Constant	0.194	0.228*	0.166	0.206			
	(1.51)	(1.80)	(1.22)	(1.53)			
UA	-0.017	-0.229**	-0.033	-0.238**			
	(-0.25)	(-2.14)	(-0.46)	(-2.09)			
Openness *UA		0.486**		0.479**			
		(2.54)		(2.32)			
Openness	0.036	-0.126	0.080	-0.096			
	(0.39)	(-1.15)	(0.78)	(-0.76)			
Disclosure	0.165***	0.170***	0.176***	0.182***			
	(3.41)	(3.57)	(3.39)	(3.57)			
Own	-0.130*	-0.087	-0.132	-0.087			
	(-1.67)	(-1.11)	(-1.60)	(-1.04)			
MarketTradingVolume	0.056	0.078	0.091	0.106			
	(0.92)	(1.30)	(1.35)	(1.59)			
MarketVolatility	0.063	0.090**	0.077	0.100**			
	(1.41)	(1.98)	(1.60)	(2.07)			
NonCatholic	-0.069	-0.017	-0.071	-0.022			
	(-1.57)	(-0.36)	(-1.52)	(-0.43)			
CivCom	0.004	-0.015	0.031	0.000			
	(0.07)	(-0.25)	(0.46)	(0.00)			
MarketDev	0.188**	0.189**	0.193**	0.200**			
	(2.40)	(2.45)	(2.29)	(2.42)			
InvProtection	-0.070	-0.188**	-0.087	-0.201**			
	(-0.93)	(-2.14)	(-1.08)	(-2.17)			
Enforcement	-0.076	-0.029	-0.097	-0.048			
<b>,</b>	(-0.93)	(-0.35)	(-1.12)	(-0.55)			
Year FEs	Yes	Yes	Yes	Yes			
N	154	154	144	144			
R2-Adj	0.225	0.256	0.225	0.252			
F	3.119	3.395	2.979	3.192			

This table shows the significance of uncertainty avoidance regarding its effect upon the relation between current stock returns and firm's future earnings. Our dependent variable is the future earnings response coefficient estimated using Equation (1) for each country-year portfolio with a minimum of 20 observations. We control for country variables at Column 1 and 3 and include an interaction term between uncertainty avoidance and market openness at Column 2 and 4. Appendix A offers detailed definitions of the variables used. T-statistics are shown in parentheses. \*\*\*, \*\* and \* indicate statistical significance at the 1%, 5% and 10% levels, respectively.